\documentclass[a4paper]{amsart}

\pdfoutput=1

\usepackage[utf8]{inputenc}
%\usepackage[french,english]{babel}
\usepackage{amsthm, amssymb, amsmath, amsfonts, mathrsfs}
\usepackage[colorlinks=true, pdfstartview=FitV, linkcolor=blue, citecolor=blue, urlcolor=blue,pagebackref=false]{hyperref}


%\usepackage{a4wide}  %\usepackage{fullpage}
%\usepackage{amscd, graphicx, epsfig, psfrag, enumerate, dsfont}

%\usepackage{color}

%\usepackage[pagewise]{lineno}
%\linenumbers

\usepackage{microtype}

%\usepackage[notcite,notref,color]{showkeys}
%\definecolor{labelkey}{gray}{.8}
%\definecolor{refkey}{gray}{.8}

\newtheorem{thm}{Theorem}[section]
\newtheorem{prop}[thm]{Proposition}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{cor}[thm]{Corollary}
\theoremstyle{remark}
\newtheorem{rem}[thm]{Remark}
%\theoremstyle{definition}


\renewcommand{\le}{\leqslant}
\renewcommand{\ge}{\geqslant}
\renewcommand{\subset}{\subseteq}
\newcommand{\mcl}{\mathcal}
\newcommand{\msf}{\mathsf}
\newcommand{\B}{\mathbb{B}}
\newcommand{\e}{\mathbf{e}}
\newcommand{\E}{\mathbb{E}}
\newcommand{\cE}{\mathcal{E}}
\renewcommand{\L}{\mathcal{L}}
\newcommand{\N}{\mathbb{N}}
\newcommand{\na}{\nabla}
\newcommand{\Ll}{\left}
\newcommand{\Rr}{\right}
\newcommand{\1}{\mathbf{1}}
\newcommand{\R}{\mathbb{R}}
\newcommand{\Q}{\mathbb{Q}}
\newcommand{\Z}{\mathbb{Z}}
%\newcommand{\z}{^{(0)}}
\newcommand{\z}{^{\circ}}
\renewcommand{\P}{\mathbb{P}}
\newcommand{\p}{^{(p)}}
\newcommand{\pp}{^{(\ov{p})}}
\newcommand{\ppp}{^{(\ov{p}+p)}}
\newcommand{\ppvp}{^{(\ov{p},p)}}
\newcommand{\ppep}{^{(\ov{p}, e^+)}}
\newcommand{\ppem}{^{(\ov{p}, e^-)}}
\newcommand{\ov}{\overline}
\newcommand{\un}{\underline}
\newcommand{\td}{\tilde}
\newcommand{\eps}{\varepsilon}
\renewcommand{\d}{{\mathrm{d}}}
\newcommand{\mclN}{\mathcal{N}}
\newcommand{\var}{\mathbb{V}\mathrm{ar}}
\newcommand{\mfk}{\mathfrak}
\newcommand{\Ah}{A_\mathrm{hom}}
\newcommand{\dr}{\partial}
\renewcommand{\o}{\omega}
\newcommand{\on}{[\un{\o}]_n}
\renewcommand{\O}{\Omega}
\newcommand{\tp}{^{(\td{p})}}
\renewcommand{\div}{\mathop{\rm div}}

\newcommand{\C}{\mathsf{C}}
\renewcommand{\H}{\mathsf{H}}

\DeclareMathOperator*{\osc}{Osc}
\DeclareMathOperator{\arsinh}{arsinh}
\DeclareMathOperator{\ent}{\msf{Ent}}

\numberwithin{equation}{section}
\renewcommand{\theequation}{\thesection .\arabic{equation}}

\title[Expansion of homogenized coefficients]{First-order expansion of homogenized coefficients under Bernoulli perturbations}
\author{Jean-Christophe Mourrat}

\address{EPFL, institut de mathématiques, station 8, 1015 Lausanne, Switzerland}

\begin{document}
\begin{thebibliography}{99}

\bibitem[Al13]{alm} Y.\ Almog. Averaging of dilute random media: a rigorous proof of the Clausius–Mossotti formula. \emph{Arch.\ Ration.\ Mech.\ Anal.}\ \textbf{207} (3), 785–812 (2013).

\bibitem[AL10a]{analeb1} A.\ Anantharaman, C.\ Le Bris. Homogénéisation d'un matériau périodique faiblement perturbé aléatoirement. \emph{C.\ R.\ Math.\ Acad.\ Sci.\ Paris} \textbf{348} (9-10), 529–534 (2010).

\bibitem[AL10b]{analeb3} A.\ Anantharaman, C.\ Le Bris. Elements of mathematical foundations for a numerical approach for weakly random homogenization problems. Preprint, arXiv:1005.3922 (2010).

\bibitem[AL11]{analeb2} A.\ Anantharaman, C.\ Le Bris. A numerical approach related to defect-type theories for some weakly random problems in homogenization. \emph{Multiscale Model. Simul.} \textbf{9} (2), 513--544 (2011).

%\bibitem[ABDH10]{andr} S.\ Andres, M.T.\ Barlow, J.-D.\ Deuschel, B.M.\ Hambly.
%Invariance principle for the random conductance model. Preprint (2010).
%

\bibitem[BFMP08]{balgar} G.\ Bal, J.\ Garnier,S.\ Motsch, V.\ Perrier. Random integrals and correctors in homogenization. \emph{Asymptot.\ Anal.}\ \textbf{59} (1-2), 1–26 (2008).

\bibitem[Ba04]{barlow} M.T.\ Barlow. Random walks on supercritical percolation clusters. \emph{Ann.\ Probab.}\ \textbf{32} (4), 3024–3084 (2004).

%\bibitem[BH09]{barham} M.T.\ Barlow, B.M.\ Hambly. Parabolic Harnack inequality and local limit theorem for percolation clusters. \emph{Electron. J. Probab.} \textbf{14} (1), 1–27 (2009). 
%

\bibitem[Be05]{bel} J.\ Beltrán. Regularity of diffusion coefficient for nearest neighbor asymmetric simple exclusion on $\Z$. \emph{Stochastic Process.\ Appl.}\ \textbf{115} (9), 1451–1474 (2005).

\bibitem[BDKY11]{benjdum} I.\ Benjamini, H.\ Duminil-Copin, G.\ Kozma, A.\ Yadin. Disorder, entropy and harmonic functions. Preprint, ArXiv:1111.4853 (2011).

%\bibitem[BLP]{blp} A.\ Bensoussan, J.-L.\ Lions, G.\ Papanicolaou. \emph{Asymptotic analysis for periodic structures}. Studies in mathematics and its applications \textbf{5}, North-Holland publishing (1978).
%
%\bibitem[BBHK08]{bbhk} N.\ Berger, M.\ Biskup, C.E.\ Hoffman, G.\ Kozma. Anomalous heat-kernel decay for random walk among bounded random conductances. \emph{Ann. Inst. Henri Poincaré Probab. Stat.} \textbf{44} (2), 374–392 (2008).
%

\bibitem[BM01]{bermit} L.\ Berlyand, V.\ Mityushev. Generalized Clausius-Mossotti formula for random composite with circular fibers. \emph{J.\ Statist.\ Phys.}\ \textbf{102} (1-2), 115–145 (2001).

\bibitem[Be02]{ber} C.\ Bernardin. Regularity of the diffusion coefficient for lattice gas reversible under Bernoulli measures. \emph{Stochastic Process.\ Appl.}\ \textbf{101} (1), 43–68 (2002). 

%\bibitem[BB10]{bbou} M.\ Biskup, O.\ Boukhadra. Subdiffusive heat-kernel decay in four-dimensional i.i.d.\ random conductance models. Preprint, 	arXiv:1010.5542v2 (2010).
%

\bibitem[BLL07]{bll} X.\ Blanc, C.\ Le Bris, P.-L.\ Lions. Stochastic homogenization and random lattices. \emph{J.\ Math.\ Pures Appl.}\ \textbf{88} 34–63 (2007).

%\bibitem[Bo82]{bo} E.\ Bolthausen. Exact convergence rates in some martingale central limit theorems. \emph{Ann. Probab.} \textbf{10} (3), 672–688 (1982). 
%

\bibitem[BP04]{boupia} A.~Bourgeat, A.~Piatnitski. Approximations of effective coefficients in stochastic homogenization. \emph{Ann.\ Inst.\ H.\ Poincar\'e Probab.\ Statist.}\ \textbf{40} (2), 153-165 (2004). 

%\bibitem[CS10]{cafsou} L.A.\ Caffarelli, P.E.\ Souganidis. Rates of convergence for the homogenization of fully nonlinear uniformly elliptic pde in random media. \emph{Invent. Math.} \textbf{180} (2), 301–360 (2010).
%

\bibitem[CI03]{capiof} P.\ Caputo, D.\ Ioffe. Finite volume approximation of the effective diffusion matrix: the case of independent bond disorder. \emph{Ann.\ Inst.\ H.\ Poincar\'e Probab.\ Statist.}\ \textbf{39} (3), 505-525 (2003). 

%\bibitem[CKS87]{cks} E.A.~Carlen, S.~Kusuoka, D.W.~Stroock. Upper bounds for symmetric Markov transition functions. \emph{Ann. Inst. Henri Poincaré Probab. Stat.} \textbf{23} (S2), 245-287 (1987).
%
\bibitem[CN00a]{connadny} J.\ Conlon, A.\ Naddaf. Green's functions for elliptic and parabolic equations with random coefficients. \emph{New York J. Math.} \textbf{6}, 153–225 (2000).
%
%\bibitem[CN00b]{connadejp} J.\ Conlon, A.\ Naddaf. On homogenization of elliptic equations with random coefficients. \emph{Electron. J. Probab.} \textbf{5} (9), 58 pp (2000).
%
%\bibitem[CS11]{conspe} J.\ Conlon, T.\ Spencer. Strong convergence to the homogenized limit of elliptic equations with random coefficients. Preprint, 	arXiv:1101.4914v1 (2011).
%

\bibitem[CK08]{cudkom}  M.\ Cudna, T.\ Komorowski. A finite dimensional approximation of the effective diffusivity for a symmetric random walk in a random environment. \emph{J.\ Comput.\ Appl.\ Math.}\ \textbf{213} (1), 186–204 (2008).

%\bibitem[DFGW89]{masi} A.~De Masi, P.A.~Ferrari, S.~Goldstein, W.D.~Wick. An invariance principle for reversible Markov processes. Applications to random motions in random environments. \emph{J. Statist. Phys.} \textbf{55} (3-4), 787-855 (1989).
%
\bibitem[De99]{del} T.\ Delmotte. Parabolic Harnack inequality and estimates of Markov chains on graphs. \emph{Rev. Mat. Iberoamericana} \textbf{15} (1), 181–232 (1999).
%
\bibitem[DD05]{deldeu} T.\ Delmotte, J.-D.\ Deuschel. On estimating the derivatives of symmetric diffusions in stationary random environment, with applications to $\nabla \phi$ interface model. \emph{Probab. Theory Related Fields} \textbf{133} (3), 358–390 (2005). 

\bibitem[DS96]{diasal} P.\ Diaconis, L.\ Saloff-Coste. Logarithmic Sobolev inequalities for finite Markov chains. \emph{Ann.\ Appl.\ Probab.}\ \textbf{6} (3), 695–750 (1996).

%
%\bibitem[Dv72]{dvo} A.\ Dvoretzky. Asymptotic normality for sums of dependent random variables. \emph{Proceedings of the sixth Berkeley symposium on mathematical statistics and probability 1970/1971}, vol.~II: Probability theory, 513–535 (1972).
%
\bibitem[EGMN12]{cemracs} A.-C.\ Egloffe, A.\ Gloria, J.-C.\ Mourrat, T.N.\ Nguyen. Random walk in random environment, corrector equation, and homogenized coefficients: from theory to numerics, back and forth. Preprint, arXiv:1211.1834.
%
%\bibitem[FM06]{fonmat} L.R.G.\ Fontes, P.\ Mathieu. On symmetric random walks with random conductances on $\Z^d$. \emph{Probab. Theory Related Fields} \textbf{134} (4), 565–602 (2006).
%

\bibitem[GNO13]{gno} A.\ Gloria, S.\ Neukamm, F.\ Otto. Quantification of ergodicity in stochastic homogenization: optimal bounds via spectral gap on Glauber dynamics -- long version. Preprint, hal-00782075 (2013).

\bibitem[GO11]{glotto} A.\ Gloria, F.\ Otto. An optimal variance estimate in stochastic homogenization of discrete elliptic equations. \emph{Ann.\ Probab.}\ \textbf{39} (3), 779–856 (2011). 

\bibitem[GO12]{glotto2} A.\ Gloria, F.\ Otto. An optimal error estimate in stochastic homogenization of discrete elliptic equations. \emph{Ann.\ Appl.\ Probab.}\ \textbf{22} (1), 1--28 (2012).

\bibitem[GZ03]{guizeg}  A.\ Guionnet, B.\ Zegarlinski. Lectures on logarithmic Sobolev inequalities. \emph{Séminaire de probabilités XXXVI} 1–134, Lecture notes in math. \textbf{1801}, Springer (2003).

%
%\bibitem[Ha84]{ha84} E.\ Haeusler. A note on the rate of convergence in the martingale central limit theorem. \emph{Ann.\ Probab.} \textbf{12} (2), 635-639 (1984).
%
%\bibitem[Ha88]{ha88} E.\ Haeusler. On the rate of convergence in the central limit theorem for martingales with discrete and continuous time. \emph{Ann.\ Probab.} \textbf{16} (1), 275-299 (1988).
%
%\bibitem[HJ88]{hj} E.\ Haeusler, K.\ Joos. A nonuniform bound on the rate of convergence in the martingale central limit theorem \emph{Ann.\ Probab.} \textbf{16} (4), 1699-1720 (1988).
%
%\bibitem[HB70]{hb70} C.C.~Heyde, B.M.~Brown. On the departure from normality of a certain class of martingales. \emph{Ann. Math. Statist.} \textbf{41}, 2161-2165 (1970).
%

\bibitem[Hu09]{hug} B.\ Hughes. Conduction and diffusion in percolating systems. \emph{Encyclopedia of complexity and systems science} 1395--1424 (2009).

%\bibitem[JS]{js} J.\ Jacod, A.N.\ Shiryaev. \emph{Limit theorems for stochastic processes} (2nd edition). Grundlehren der mathematischen Wissenschaften \textbf{288}. Springer (2003).
%
%\bibitem[JKO]{jko} V.V.~Jikov, S.M.~Kozlov, O.A.~Oleinik. \emph{Homogenization of differential operators and integral functionals}. Translated from Russian by G.A.\ Yosifian. Springer (1994).
%
%\bibitem[Ka]{kato} T.\ Kato. \emph{Perturbation theory for linear operators} (2nd edition). Grundlehren der mathematischen Wissenschaften \textbf{132}, Springer (1976).
%
\bibitem[KV86]{kipvar} C.\ Kipnis, S.R.S.\ Varadhan. Central limit theorem for additive functionals of reversible Markov processes and applications to simple exclusions, \emph{Comm. Math. Phys.} \textbf{104}, 1-19 (1986).
%

\bibitem[Ko89]{koz} S.M.\ Kozlov. Geometric aspects of averaging. (Russian) \emph{Uspekhi Mat.\ Nauk} \textbf{44} (2), 79--120 (1989). Transl.\ in \emph{Russian Math.\ Surveys} \textbf{44} (2), 91--144 (1989).

\bibitem[KO05]{komoll} T.\ Komorowski, S.\ Olla. Einstein relation for random walks in random environments. \emph{Stochastic Process.\ Appl.}\ \textbf{115} (8), 1279–1301 (2005).

%\bibitem[Ko78]{kozlov1} S.M.~Kozlov. Averaging of random structures. \emph{Dokl. Akad. Nauk SSSR} \textbf{241} (5), 1016-1019 (1978). English transl.~: \emph{Soviet Math. Dokl.} \textbf{19} (4), 950-954  (1978).
%
\bibitem[Kü83]{kuen} R.\ Künnemann. The diffusion limit for reversible jump processes on $\Z^d$ with ergodic random bond conductivities. \emph{Comm. Math. Phys.} \textbf{90} (1), 27–68 (1983). 
%

\bibitem[LOV01]{lov} C.\ Landim, S.\ Olla, S.R.S.\ Varadhan. Symmetric simple exclusion process: regularity of the self-diffusion coefficient. \emph{Comm.\ Math.\ Phys.}\ \textbf{224} (1), 307–321 (2001).

\bibitem[LOV04]{lov2} C.\ Landim, S.\ Olla, S.R.S.\ Varadhan.
On viscosity and fluctuation-dissipation in exclusion processes. \emph{J.\ Statist.\ Phys.}\ \textbf{115} (1-2), 323–363 (2004). 

%\bibitem[Le01]{lejay} A.\ Lejay. Homogenization of divergence-form operators with lower-order terms in random media. \emph{Probab. Theory Related Fields} \textbf{120} (2), 255–276 (2001). 
%

\bibitem[MO13]{marott} D.\ Marahrens, F.\ Otto. Annealed estimates on the Green's function. Preprint, arXiv:1304.4408v1 (2013).


\bibitem[Ma]{maxwell} J.C.~Maxwell. Medium in which small spheres are uniformly disseminated. \emph{A treatise on electricity and magnetism}, 3d ed., part II, chapter IX, article 314. Clarendon press (1891).
%
%\bibitem[Mo11]{berry} J.-C.\ Mourrat. A quantitative central limit theorem for the random walk among random conductances. Preprint, arXiv:1105.4485v1.
%
%\bibitem[Mo12]{martingale_CLT} J.-C.\ Mourrat. On the rate of convergence in the martingale central limit theorem. \emph{Bernoulli}, to appear.
%
\bibitem[Mo11]{vardecay} J.-C.\ Mourrat. Variance decay for functionals of the environment viewed by the particle. \emph{Ann.\ Inst.\ H.\ Poincaré Probab.\ Statist.}\ \textbf{47} (1), 294-327 (2011).

\bibitem[Na05]{nag1} Y.\ Nagahata. Regularity of the diffusion coefficient matrix for the lattice gas with energy. \emph{Ann.\ Inst.\ H.\ Poincaré Probab.\ Statist.}\ \textbf{41} (1), 45–67 (2005).

\bibitem[Na06]{nag2} Y.\ Nagahata. Regularity of the diffusion coefficient matrix for generalized exclusion process. \emph{Stochastic Process.\ Appl.}\ \textbf{116} (6), 957–982 (2006).

\bibitem[Na07]{nag3} Y.\ Nagahata. Regularity of the diffusion coefficient matrix for lattice gas reversible under Gibbs measures with mixing condition. \emph{Comm.\ Math.\ Phys.}\ \textbf{273} (3), 637–650 (2007).

\bibitem[Pa95]{pap} G.C.~Papanicolaou. Diffusion in random media. \emph{Surveys in applied mathematics} 205--255 (J.B.\ Keller et al.\ Eds), Plenum Press (1995).

\bibitem[PV81]{papvar} G.C.~Papanicolaou, S.R.S.~Varadhan. Boundary value problems with rapidly oscillating random coefficients. \emph{Random fields (Esztergom, 1979)} 835-873, Colloq. Math. Soc. J\'anos Bolyai \textbf{27}, North-Holland (1981). 
%
%\bibitem[RR]{raru} S.T.\ Rachev, L.\ Rüschendorf. \emph{Mass transportation problems -- Vol.\ II: applications}. Probability and its applications, Springer (1998).
%
%\bibitem[RS]{rs-fourier} M.\ Reed, B.\ Simon. \emph{Methods of modern mathematical physics II -- Fourier analysis, self-adjointness}. Academic press (1975).

\bibitem[SC]{sc} L.\ Saloff-Coste. Lectures on finite Markov chains. \emph{Lectures on probability theory and statistics (Saint-Flour 1996)} 301–413, Springer (1997). 

%
%\bibitem[SS04]{sid} V.\ Sidoravicius, A.-S.\ Sznitman. Quenched invariance principles for walks on clusters of percolation or among random conductances. \emph{Probab. Theory Related Fields} \textbf{129} (2), 219-244 (2004).
%

\bibitem[SW01]{smwe} S.\ Smirnov, W.\ Werner. Critical exponents for two-dimensional percolation. \emph{Math.\ Res.\ Lett.}\ \textbf{8} (5-6), 729–744 (2001).

\bibitem[SZ97]{stzh} D.W.\ Stroock, W.\ Zheng. Markov chain approximations to symmetric diffusions. \emph{Ann.\ Inst.\ H.\ Poincaré Probab.\ Statist.}\ \textbf{33} (5), 619–649 (1997). 

\bibitem[St]{ray} J.W.~Strutt (3d Baron Rayleigh). On the influence of obstacles arranged in rectangular order upon the properties of a medium. \emph{Philos. mag.} \textbf{34}, 481-502 (1892).

\bibitem[Su05]{sue} M.\ Sued. Regularity properties of the diffusion coefficient for a mean zero exclusion process. \emph{Ann.\ Inst.\ H.\ Poincaré Probab.\ Statist.}\ \textbf{41} (1), 1–33 (2005).

%
%\bibitem[Vi]{villani} C.\ Villani. \emph{Topics in optimal transportation.}
%Graduate studies in mathematics \textbf{58}, American mathematical society (2003).
%
%\bibitem[Yu80]{yuri1} V.V.~Yurinski\u{\i}. On a Dirichlet problem with random coefficients. \emph{Stochastic differential systems (Proc. IFIP-WG 7/1 Working Conf., Vilnius, 1978)} 344-353, Lecture Notes in Control and Information Sci. \textbf{25}, Springer (1980).
%
%\bibitem[Yu86]{yuri} V.V.~Yurinski\u{\i}. Averaging of symmetric diffusion in a random medium (in Russian). \emph{Sibirsk. Mat. Zh.} \textbf{27} (4), 167-180 (1986). English transl. in \emph{Siberian Math. J.} \textbf{27} (4), 603-613 (1986).

\bibitem[ZB77]{zuzbre} M.\ Zuzovsky, H.\ Brenner. Effective conductivities of composite materials composed of cubic arrangements of spherical particles embedded in an isotropic matrix. \emph{Journal of applied mathematics and physics (ZAMP)} \textbf{28}, 979--992 (1977).




\end{thebibliography}

%

\begin{abstract}

Divergence-form operators with stationary random coefficients homogenize over large scales. We investigate the effect of certain perturbations of the medium on the homogenized coefficients. The perturbations considered are rare at the local level, but when occurring, have an effect of the same order of magnitude as the initial medium itself. The main result of the paper is a first-order expansion of the homogenized coefficients, as a function of the perturbation parameter. 

\bigskip

\noindent \textsc{MSC 2010:} 35B27, 35J15, 35R60, 82D30.

\medskip

\noindent \textsc{Keywords:} homogenization, random media, Clausius-Mossotti formula, Bernoulli perturbation.

\end{abstract}
%
\maketitle
%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Introduction}
\label{s:intro}
%\setcounter{equation}{0}

Consider a divergence-form operator whose coefficients are random. The randomness of the coefficients model the small-scale irregularity of a medium. If the distribution of the coefficients is stationary and ergodic (and if some ellipticity condition holds), then this random operator can, over large scales, be replaced by an effective operator with constant \emph{homogenized} coefficients. %This homogenization property was already understood from a physical point of view in the late 19th century \cite{max,ray}, and started to receive a mathematically rigorous treatment in the 60's and 70's \cite{jko}.

%The homogenized coefficients are given in terms of the \emph{corrector}. 

The aim of this paper is to study the effect of certain perturbations of the medium on the homogenized coefficients. The perturbations considered are small only in the sense that locally, the medium is perturbed with small probability; but where a perturbation occurs, the change is of the same order of magnitude as the medium itself. This type of perturbation may be called a \emph{Bernoulli perturbation}. Our main purpose is to prove a first-order expansion of the homogenized coefficients, as a function of the perturbation parameter, under conditions of short-range correlations and uniform ellipticity of the medium.

\medskip 

There are at least two important motivations behind this problem. The first concerns the numerical approximation of the homogenized coefficients. Although several techniques for doing so have been identified and analysed \cite{cemracs}, it remains a computationally expensive task, even in low dimensions. It has thus been proposed in \cite{analeb1} to study more efficient techniques that would apply to \emph{weakly random} media, that is, random perturbations of a periodic environment. First-order expansions (as a function of the perturbation parameter) have been proved in \cite{analeb3}, but only for specific types of perturbations that do not include Bernoulli perturbations. Yet, Bernoulli perturbations are arguably the most natural modelling assumption for typical disordered media like composite materials (as an example, see \cite{analeb2} for a cross-section of a composite material used in the aeronautics industry). In \cite{analeb2,analeb3}, conjectures are formulated concerning the expansion of the homogenized coefficients for such perturbations, which are backed by a formal derivation and numerical evidence. 

A second motivation is related to percolation. For this model, each edge of $\Z^d$ is independently removed with probability $p$, and kept otherwise. There exists a critical probability $p_c \in (0,1)$ such that the remaining graph has a unique infinite connected component if $p < p_c$, and has only finite connected components if $p > p_c$. Over the last years, the understanding of two-dimensional percolation close to criticality progressed tremendously, in particular through the rigorous derivation of the values of several critical exponents (see for instance \cite{smwe}). Yet, to the best of my knowledge, it is not known (for any $d \ge 2$) how the homogenized conductivity of the percolation cluster behaves as $p$ approaches $p_c$ from below (see \cite{hug} for a nice review of the problem). Understanding the effect of small perturbations of the medium on the homogenized conductivity seems to be a first necessary step towards the resolution of this problem.

\medskip 

Of the two motivations above, the first is formulated for differential operators, while the second is inherently discrete. In what follows, we focus on a discrete model. Apart from this difference, the results presented here give a proof of the first-order expansion of the homogenized coefficients conjectured in \cite{analeb1, analeb2, analeb3}. It extends it in the sense that the non-perturbed environment considered here need not be periodic, and that the expansion is obtained around every value of the perturbation parameter. 
We work under a condition of uniform ellipticity, which is obviously not satisfied in the case of percolation. The present paper will hopefully lay the basis for an extension to the case of percolation, as well as to higher-order expansions.

\medskip

On a heuristic level, the first-order expansion of the homogenized coefficients can be guessed as follows. The homogenized coefficients can be expressed in terms of the \emph{corrector}, whose defining equation is posed on the whole space. It is well-known that finite-volume approximations of the corrector yield consistent approximations as the volume tends to infinity. One can easily derive a first-order expansion of the corrector defined over a given finite volume element (or any reasonable function of it) as the perturbation parameter tends to $0$. Indeed, in this limit, one can assume at first order that there is at most one location that is perturbed. One then gets a formula for the first-order expansion of the homogenized coefficients by formally interchanging the ``infinite-volume'' and the ``small perturbation parameter'' limits.

We will see here that this informal reasoning can be made rigorous, and thus provide us with a first-order expansion of the homogenized coefficients. The main point is to quantify errors when localizing the problem over a finite volume, say of side length~$N$, and then choose $N$ as a function of the perturbation parameter $p$. It is clear that, for this strategy to make sense at all, we need the box to contain some perturbed locations, so we should have $N^d \gg p^{-1}$. A more refined heuristic consists in observing that a random walk evolving in a box of size $N$ sees only of order $N^2$ sites, so that we should in fact need $N^2 \gg p^{-1}$. Using parabolic/elliptic regularity theory, the averaged estimates on the gradients of the Green function of \cite{deldeu}, and the localization error estimates on the corrector due to \cite{glotto, glotto2}, we will see that the argument can be made rigorous if we choose $N^2 = p^{-(1+\eps)}$ for some small $\eps > 0$. (One should think of $N$ as being $\mu^{-1/2}$ in the notation introduced below.) 

Of these three main ingredients, only the estimates of \cite{glotto,glotto2} require the assumption on the short range of correlations. At bottom, what is required for these estimates to hold is a spectral gap inequality for the Glauber dynamics, which makes so-called ``vertical derivatives'' come in (see \cite{gno} and Remark~\ref{r:corr} below for more on this). %The condition that the medium satisfies this spectral gap inequality is more general than the assumption 

In order to extend the present results to the case of continuous operators, the only missing main ingredient is an analogue of the results of \cite{glotto, glotto2}. Results in this direction have been announced by the authors. Concerning the extension to percolation, it is noteworthy that a probabilistic version of the Harnack inequality was proved in \cite{barlow}, while part of the estimates of \cite{deldeu} have been extended to percolation clusters in \cite{benjdum}. 

In our discrete, uniformly elliptic setting, stronger results on the Green function and its gradients have been obtained very recently in \cite{marott}. We will see in the last section how to use them to get a sharp control of the error term in the expansion.



\medskip

Let us now give a brief survey of related works, outside of the previously mentioned \cite{analeb1,analeb2,analeb3}. 


In some cases, asymptotic formulas for the homogenized coefficients in the regime of high dilution of the perturbations have been known and used by physicists since the 19th century, and are variously known as the Clausius-Mossotti, Lorentz-Lorenz, Maxwell, or Rayleigh formulas. The setting is that of a continuous and homogeneous medium with highly diluted spherical inclusions. In this context, the general expression for the first-order correction that will be obtained here in terms of correctors becomes an explicit formula \cite{maxwell,pap}. Vanishingly small spherical inclusions arranged periodically along a fixed square or cubic lattice within a homogeneous medium have been investigated with precision in \cite{ray}, where a high-order expansion is obtained. Surprisingly, in dimension 3, the expansion involves fractional powers of the volume fraction $p$ of the small spherical inclusions, the smallest fractional power with a non-zero coefficient being $p^{4+1/3}$ (see \cite[(60) and (64)]{ray}). In particular, the homogenized coefficients can be differentiated four times with respect to $p$, but no more. Note that in \cite{ray}, every periodic cell is perturbed by the insertion of a small inclusion, while our present approach would cover the different situation where each periodic cell is perturbed by some fixed inclusion, with small probability. This difference should be irrelevant as far as the first order of the expansion is concerned, but not so for higher orders. In our context of random perturbations, it is an open question whether the homogenized coefficients are infinitely differentiable functions of the perturbation parameter. 

Clausius-Mossotti formulas have been the subject of an impressive amount of work in the physics literature, of which we simply quote \cite{zuzbre}, where results of \cite{ray} are extended to higher-order corrections. In the mathematics literature, the most important result to date on this problem (that I know of) is certainly \cite{alm}. Under some conditions, the Clausius-Mossotti formula is proved (although the results obtained there do not directly relate to the homogenized coefficients), with an error bound of order $p^{3/2}$, where $p$ is the volume fraction of the perturbation. In our context, we will see that the first-order expansion of the homogenized coefficients holds with an error term of order $o(p^{2-\eta})$, for every $\eta > 0$. Earlier investigations include \cite{koz,bermit}.



The problematic considered in \cite{bll} is similar in spirit to the present one. In this work, a different type of perturbation, based on random deformations of the geometry of the medium, has been investigated, and a first-order expansion has been obtained, see \cite[Theorem~3.2]{bll}.



The problem of showing the regularity of diffusion coefficients has also been explored for others models, and in particular for that of a tagged particle in the simple symmetric exclusion process. It is proved in \cite{lov} that the effective diffusivity for this model is an infinitely differentiable function of the density of particles (see also \cite{ber,lov2,bel,sue,nag1,nag2,nag3} for generalizations). The approach followed there relies on a particular duality structure of the process. In our present context, this duality has been investigated in \cite{komoll,cudkom} under the additional assumption that the random coefficients take only two possible values (although no regularity result on the diffusion coefficients was given there). In contrast, the approach of the present paper does not involve any duality structure, and (as a consequence) is not restricted to random coefficients taking only a finite number of possible values. 


%The problem of showing the regularity of diffusion coefficients has already been explored in the literature for others models, and in particular for that of a tagged particle in the simple symmetric exclusion process. It is proved in \cite{lov} that the effective diffusivity for this model is an infinitely differentiable function of the density of particles (see also \cite{ber,lov2,bel,sue,nag1,nag2,nag3} for generalizations). The approach taken up there relies on a particular duality structure of the process, which does not seem to be applicable to our present context (see however \cite{komoll,cudkom} for an exploration of this duality for random coefficients taking only two possible values). In contrast, the approach of the present paper does not involve any duality structure, but only differentiability is proved. Although a generalization of the ideas presented here might yield higher-order expansions, a proof of infinite differentiability of the homogenized coefficients seems to require substantially different ideas.


\medskip

In the next section, we introduce some definitions and notations, and then explain in more detail the approach taken up and the organization of the rest of the paper.

%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Definitions and notations}
\label{s:defs}

\subsection{A reminder on homogenization}
We view $\Z^d$ ($d \ge 2$) as a graph, with its usual nearest-neighbour structure, and write $\B$ for the set of (non-oriented) edges. We write $x\sim y$ if $x$ and $y \in \Z^d$ are neighbours.
%
%
Let $(\omega_e)_{e \in \B}$ be i.i.d.\ random variables such that almost surely, 
$$
c_- \le \omega_e \le c_+ ,
$$ 
where $0 < c_- < c_+ < +\infty$ are constants, henceforth called the \emph{ellipticity constants}. We call $\omega = (\omega_e)_{e \in \B}$ an \emph{environment}, and write $\Omega = [c_-,c_+]^{\B}$ for the set of environments. We call $\omega_e$ the \emph{conductance} of the edge $e$. We write $\P$ for the law of~$\omega$, and $\E$ for the associated expectation. Note that $\Z^d$ acts on $\Omega$ by the translations $(\theta_z)_{z \in \Z^d}$ defined by $(\theta_z \ \omega)_{x,y} = \omega_{x+z,y+z}$.

Let $(\e_1, \ldots, \e_d)$ be the canonical basis of $\R^d$. For every $x \in \Z^d$, let $A(x,\omega) = A^\o(x)$ be the $d$-dimensional diagonal matrix
$\textrm{diag}(\omega_{x,x+\e_1}, \ldots,\omega_{x,x+\e_d})$. We may keep the variable $\omega$ implicit in the notation. The operator whose homogenization properties are investigated here is $-\na^* \cdot A \na$, which acts on functions $f : \Z^d \to \R$ by
$$
-\na^* \cdot A \na f (x) = \sum_{y \sim x} \omega_{x,y} (f(y)-f(x)) \qquad (x \in \Z^d).
$$
Here and below, we write $\na f$ to denote the forward gradient,
$$
\nabla f(x)=\left[  
\begin{array}{c}
f(x+\e_1)-f(x) \\
\vdots\\
f(x+\e_d)-f(x)
\end{array}
\right],
$$
and for $F = (F_1,\ldots,F_d): \Z^d \to \R^d$, we write $\na^* \cdot F$ for the backward divergence, 
$$
\na^* \cdot F(x) = \sum_{i = 1}^d \Ll(F_i(x) - F_i(x-\e_i)\Rr).
$$

Homogenization refers to the fact that there exists a \emph{constant} symmetric matrix $\Ah$ such that the solution operator of $-\na^* \cdot A(\cdot/\eps) \na$ converges, as $\eps \to 0$, to the solution operator of the differential operator $-\na \cdot \Ah \na$. Moreover, the homogenized matrix $\Ah$ can be characterized in terms of a function called the corrector, which we now proceed to define. 

We say that a function $\psi : \Z^d \times \Omega \to \R$ is \emph{stationary} if $\psi(x,\omega) = \psi(0,\theta_x \ \omega)$. Let $\xi \in \R^d$ be a vector which will be kept fixed throughout this paper.  The \emph{corrector} $\phi$ in the direction $\xi$ is the unique function: $\Z^d \times \Omega \to \R$ such that $\na \phi$ is stationary, $\E[\na \phi] = 0$, $\phi(0) = 0$ and 
\begin{equation}
\label{defphip}
-\na^* \cdot A(\xi + \na \phi)(x,\omega) = 0 \qquad (x \in \Z^d,\  \P\text{-a.e. } \omega).
\end{equation}
There are several ways to define the homogenized matrix in terms of the corrector, but the most useful one for our present purpose is the property that
\begin{equation}
\label{defAh}
\xi \cdot \Ah \xi = \E[\xi \cdot A(\xi + \na \phi)].
\end{equation}
In the right-hand side above, we keep implicit the fact that the quantity under the expectation is evaluated at $(0,\omega)$. %Note that by stationarity of $\na \phi$ and the fact that the measure $\P$ is translation invariant, we have
%$$
%\E[\na \phi(x,\omega)] = \E[\na \phi(0,\omega)].
%$$
%
Proofs of existence and uniqueness of $\phi$ as stated above, and of the formula \eqref{defAh}, can be found for instance in \cite[Theorem 3 and (3.17)]{kuen}, %\cite[Subsection~2.2]{cemracs}, 
or in the continuous setting, in \cite[Theorem~2]{papvar}.

\subsection{Bernoulli perturbations}
We now introduce Bernoulli perturbations. We give ourselves a second family $({\o}^{(1)}_e)_{e \in \B}$ of i.i.d.\ random variables such that almost surely,
$$
c_- \le {\o}^{(1)}_e \le c_+.
$$
For every $p \in [0,1]$, we also give ourselves a family $(b\p_e)_{e \in \B}$ of independent Bernoulli random variables of parameter $p$, independent of $(\omega_e,{\o}^{(1)}_e)_{e \in \B}$. We now understand that $\P$ denotes the joint law of $\un{\o} := (\omega_e,{\o}^{(1)}_e,(b\p_e)_{p \in [0,1]})_{e \in \B}$, so that for instance, we have $p = \P[b_e\p = 1]$. We write $\un{\Omega}$ for the set of possible values taken by~$\un{\o}$. The group $\Z^d$ acts on $\un{\Omega}$ by translations, and we keep writing $(\theta_x)_{x \in \Z^d}$ to denote this action. 

We let $\omega\p_e = (1-b\p_e)\ \omega_e + b\p_e \ {\o}^{(1)}_e$. Note that the notation is consistent when $p = 1$, and that if we denote the law of $\o\p_e$ by $\nu\p$, then $\nu\p = (1-p) \nu^{(0)} + p \nu^{(1)}$. With each $p \in [0,1]$ is thus associated the perturbed environment $\omega\p = (\omega\p_e)_{e \in \B}$. The environment $\omega\p$ shares the same properties as the environment $\omega$, and thus we can define $A\p, \Ah\p$ and $\phi\p$ in the same way as $A, \Ah$, and $\phi$ respectively, but with $\omega$ replaced by $\omega\p$. Throughout this article, for notational convenience, we will replace the exponent $^{(0)}$ by simply $^\circ$, so that for instance, $A\z = A^{(0)}$, $\Ah\z = \Ah^{(0)}$, $\phi\z = \phi^{(0)}$, and so on (we think of $A\z$, $\phi\z$, etc.\ as functions of $\un{\o}$, which makes them formally different from $A$, $\phi$, etc.\ introduced before). 

\medskip

Our main goal is to show that the homogenized matrix $\Ah\p$, as a function of~$p$, is differentiable, and to find an explicit formula for the derivative.
%
%
%
Heuristically, one may expect that a linear approximation in \eqref{defAh} gives the correct first-order approximation, that is, as $p$ tends to $0$,
\begin{equation}
\label{e:linearapprox}
\xi \cdot \Ah\p \xi = \xi \cdot \Ah\z \xi + p \sum_{e \in \B} \E[\xi \cdot A^e(\xi + \na \phi^e) - \xi \cdot A\z(\xi + \na \phi\z)] + o(p),
\end{equation}
where $A^e$ and $\phi^e$ are for the environment perturbed at the edge $e$ what $A$ and $\phi$ are for the unperturbed environment. In the expectation above and throughout this paper, it is kept implicit that the functions are evaluated at $(0,\un{\o})$. Our aim is to justify this heuristic. Note that the meaning of the sum in \eqref{e:linearapprox} is not clear, since it is not absolutely summable. 

\medskip

\noindent \textbf{Organization of the paper.} In section~\ref{s:green}, we recall classical results on the decay at infinity of the Green function and its gradients. A difficulty with the formula \eqref{defAh} is that the corrector is a very non-local function. It is convenient to introduce a localized version of the corrector, obtained by adding a zero-order term in equation~\eqref{defphip}. In section~\ref{s:solve}, we give a simple criterion for the existence and uniqueness of solutions of elliptic equations with a zero-order term. In section~\ref{s:loc}, we recall results quantifying the accuracy of the approximation by this localized corrector. Although the (localized) corrector depends non-linearly on the values taken by the environment~$\o\p$, we will show in section~\ref{s:linear} that it is, at first order, close to its linear approximation in the limit of small~$p$. After having derived several useful convergence results related to stationary elliptic equations in section~\ref{s:station}, we state and prove Theorem~\ref{t:main0} in section~\ref{s:expansion0}, which is a rigorous version of \eqref{e:linearapprox}. We show in section~\ref{s:periodiz} that localization by a zero-order term in \eqref{defphip} can be replaced by periodization of the medium, thus providing us with alternative descriptions of the derivative of $\Ah\p$ at $p = 0$. The goal of the last two sections is to generalize Theorem~\ref{t:main0} in two directions. In section~\ref{s:expansion}, we give the asymptotic expansion of $\Ah\ppp$ as $p$ tends to $0$, for every $\ov{p} \in [0,1]$. Finally, using the recent results of \cite{marott}, we prove in section~\ref{s:error} that the error term in the expansion is $o(p^{2-\eta})$ for every $\eta > 0$.


%\medskip 

\begin{rem}
\label{r:corr}
Although we work throughout under the assumption that the medium is made of i.i.d.\ random variables, the method can be generalized to some (stationary) weakly correlated environments. For this reason, we will not use some properties specific to the i.i.d.\ case (as e.g.\ the fact that the homogenized matrices $\Ah\p$ are multiples of the identity). The bottleneck in generalizing the present approach to more general distributions lies with the results of \cite{glotto,glotto2} recalled in section~\ref{s:loc}. In \cite{gno}, these results were shown to hold as soon as the law of the medium satisfies a spectral gap inequality with respect to the Glauber dynamics. This last condition is more general than the i.i.d.\ assumption, although very far from covering arbitrary ergodic media. The estimates of \cite{glotto,glotto2} are not expected to hold for arbitrary ergodic media however. This can be checked directly in dimension~$1$, or can be derived from the precise quantitative results of \cite{balgar} on homogenization in very correlated environments. (Outside of this remark, recall that we always assume $d \ge 2$ here.)
\end{rem}

%\medskip

\noindent \textbf{Notation.} We define $a \vee b=\max(a,b)$. We write $|\cdot|$ for the $L^2$ norm on $\R^d$. This norm induces an operator norm on $d \times d$ matrices, which we also denote by $|\cdot|$. For a function of several variables, e.g.\ $f : \R \times \Z^d \times \Z^d \to \R$, we write $\na_2 f$ and $\na_3 f$ to denote the (forward) gradient with respect to the second and to the third variable respectively. For $i,j \in \{2,3\}$, we write $\na_i \na_j f$ to denote the $d \times d$ matrix whose columns are 
$
\na_i F_1, \ldots, \na_i F_d,
$
where $[F_1, \ldots, F_d]^{\msf{T}} = \na_j f$. Note that if $v \in \R^d$ is a fixed vector, and $h = \na_j f \cdot v$, then $\na_i h = (\na_i \na_j f) v$. 

The value of a constant denoted by $c$ or $\td{c}$ may change from one occurrence to another, but depends only on the dimension and the ellipticity constants.
%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Parabolic and elliptic regularity}
\label{s:green}
%\setcounter{equation}{0}

%\begin{prop}
%\label{parabolic}
%For every fixed $\omega \in \Omega$, and $x \in \Z^d$, let $(p^\o(x,y,t))_{(y,t) \in \Z^d \times \R_+}$ be the unique bounded function satisfying
%$$
%\left\{
%\begin{array}{l}
%\dr_t p^\omega(x,y,t) = -\na^* \cdot A(y,\omega) \na p^\omega(x,y,t) \qquad ((y,t) \in \Z^d \times \R_+), \\
%p^\omega(x,y,0) = \1_{x = y} \qquad (y \in \Z^d).
%\end{array}
%\right.
%$$
%Let $p^*(x,y,t)$ be the corresponding function associated with the environment constant equal to $1$, and $p^*(x,t) = p^*(0,x,t)$. There exist constants $C, k, \alpha > 0$ such that, uniformly over $\omega \in \Omega$ and $x,y \in \Z^d$,
%\begin{equation}
%\label{parabpoint}
%p\omega(x,y,t) \le C p^*(x,y,kt),
%\end{equation}
%\begin{equation}
%\label{parabpoint}
%|\na_y p\omega(x,y,t) \le C p^*(x,y,kt),
%\end{equation}
%\end{prop}

The aim of this section is to gather known results on the regularity of the heat kernel and Green function associated with divergence-form operators. We begin by defining the heat kernel. For every fixed $\omega \in \Omega$ and $x \in \Z^d$, let $(q^\o(t,x,y))_{(t,y) \in \R_+ \times \Z^d}$ be the unique bounded function satisfying
$$
\left\{
\begin{array}{l}
\dr_t q^\omega(t,x,y) = -\na^* \cdot A^\o(y) \na q^\omega(t,x,y) \qquad ((t,y) \in \R_+ \times \Z^d), \\
q^\omega(0,x,y) = \1_{y = x} \qquad (y \in \Z^d),
\end{array}
\right.
$$
where we understand $\na^*$ and $\na$ as acting on the $y$ variable. Note that $q^\o(t,x,y) = q^\o(t,y,x)$. Let $q^*(t,x) = q^{\omega^*}(t,0,x)$, where $\omega^*$ is the constant environment such that $\omega^*_e = 1$ for every $e \in \B$. 

\begin{prop}[pointwise control of the heat kernel \cite{stzh,del}]
\label{parabolic}
There exist constants $c, k > 0, \alpha > 0$ such that for every $\omega \in \Omega$, $t \ge 0$, $x \in \Z^d$ and $i \in \{2,3\}$,
\begin{equation}
\label{parabpoint}
q^\omega(t,0,x) \le c \ q^*(kt,x),
\end{equation}
\begin{equation}
\label{parabpointgrad}
\Ll|\na_i q^\omega(t,0,x)\Rr| \le c \  \frac{q^*(k t,x)}{(1 \vee t)^\alpha}.
\end{equation}
\end{prop}
\begin{proof}
The first part appears in this form in \cite[Section~4]{deldeu}, and the proof given there relies on the closely related \cite[Proposition~3.4]{del} (a similar statement was also obtained in \cite[Lemma~1.9]{stzh}). It also appears that as soon as \eqref{parabpoint} holds for some value of $k$, it holds for every larger value as well (with a different $c$). This will also be clear from the proof of \eqref{parabpointgrad}, to which we now turn.

Inequality \eqref{parabpointgrad} roughly corresponds to \cite[Theorem~1.31]{stzh}. In order to provide the reader with a precise proof, we begin by recalling the following consequence of the parabolic Harnack inequality. For $r > 0$ and $x \in \Z^d$, let
$$
Q(x,r) = [0,4r^2] \times \{ z \in \Z^d : |z-x| \le 2r \},
$$
$$
Q'(x,r) = [3r^2,4r^2] \times \{z \in \Z^d : |z-x| \le r \}.
$$
For a real function $u$ defined on a set $A$, we write
$$
\osc_A u = \sup_A u - \inf_A u.
$$
\begin{prop}[increase of oscillations \cite{stzh}]
\label{p:oscill}
There exists $\lambda > 1$ such that for every $\omega \in \Omega$, $r > 0$ and $x \in \Z^d$, if $u : \R_+ \times \Z^d \to \R$ satisfies
$$
\dr_t u = -\na^* \cdot A(\cdot, \omega) \na u \qquad \text{in } Q(x,r),
$$
then
$$
\osc_{Q(x,r)} u \ge \lambda \osc_{Q'(x,r)} u.
$$
\end{prop}
The proof of Proposition~\ref{p:oscill} can be found in \cite[Lemma~1.30]{stzh}, or can be derived from the parabolic Harnack inequality given in \cite[Theorem~2.1]{del} as in the proof of \cite[Proposition~3.4]{deldeu} (see also \cite[p.\ 188]{del} on the fact that numerical constants appearing in the definitions of $Q$ and $Q'$ are irrelevant, and can thus be chosen as convenient).

We will also need the following result, which we borrow from \cite[Section~4]{deldeu}. The expressions involved are less simple than what one might imagine a priori, due to lattice effects.
\begin{prop}[explicit estimates on $q^*$]
\label{p:estimqetoile}
Let 
$$
\mcl{D}_t(x) = |x| \arsinh\Ll( \frac{|x|}{t} \Rr) + t \Ll( \sqrt{1+\frac{|x|^2}{t^2}} - 1 \Rr).
$$
There exist constants $c_1,c_2,k_1,k_2$ such that for every $t \ge 0$ and $x \in \Z^d$,
$$
\frac{c_1}{(1\vee t)^{d/2}}\exp\Ll(-\mcl{D}_{k_1t}(x)\Rr) \le q^*(t,x) \le \frac{c_2}{(1\vee t)^{d/2}}\exp\Ll(-\mcl{D}_{k_2t}(x)\Rr).
$$
\end{prop}
\begin{rem}
The following alternative expression for $\mcl{D}_t(x)$ is also given in \cite{deldeu}:
$$
\mcl{D}_t(x) = \max_{s} \Ll(s |x| - t (\cosh(s) - 1)\Rr),
$$
which makes it transparent that $t \mapsto \mcl{D}_t(x)$ is decreasing (in the wide sense).
\end{rem}
Let us see how to use these results to prove that \eqref{parabpointgrad} holds. We fix $k$ such that \eqref{parabpoint} holds, and $k_1, k_2$ as given by Proposition \ref{p:estimqetoile}. We let $k_3 > 0$ be some parameter yet to be fixed, and begin by proving that \eqref{parabpointgrad} holds provided $|x| \le k_3 t$. (This is the most relevant case, but lattice effects prevent us from giving a unified proof.)



It follows from Proposition~\ref{p:estimqetoile} that there exists constants $c_1,c_2$ such that if $|x|\le 2 k_3 k t$, then 
\begin{equation}
\label{gaussbounds}
\frac{c_1}{(1 \vee t)^{d/2}} e^{-|x|^2/c_1 t} \le q^*(t,x) \le \frac{c_2}{(1 \vee t)^{d/2}} e^{- |x|^2/c_2 t}.
\end{equation}
Let $x,t$ satisfy $|x| \le k_3 t$, and let $L = \lfloor \log_2(t/2)/2 \rfloor$. For $L < 0$ (i.e.\ $t < 2$), there is nothing to prove since \eqref{parabpoint} holds. Otherwise, for every $l \in \{0, \ldots, L\}$, we consider the cylinders
$$
Q_l =[t-2^{2l},t] \times \{z \in \Z^d : |z-x| \le 2^l\}.
$$
Let $y$ be a neighbour of $x$. Noting that $(t,x)$ and $(t,y)$ both belong to $Q_0$ and applying Proposition~\ref{p:oscill} iteratively, we obtain that
$$
\Ll| q^\omega(t,0,x) - q^\o(t,0,y) \Rr| \le \lambda^{-L} \osc_{Q_L} q^\o(\cdot,0,\cdot) \le \lambda^{-L} \sup_{Q_L} q^\o(\cdot,0,\cdot).
$$
By the definition of $L$, the cylinder $Q_L$ is included in the cylinder
$$
Q = [t/2,t] \times \{z \in \Z^d : |z-x| \le \sqrt{t}\}.
$$
In view of \eqref{parabpoint} and \eqref{gaussbounds}, we thus get that
$$
\Ll| q^\omega(t,0,x) - q^\o(t,0,y) \Rr| \le c \lambda^{-L} \ q^*(k_0 t,x),
$$
for every $k_0 > 0$ large enough, uniformly over $x$ and $t$ satisfying $|x| \le k_3 t$. This is indeed the inequality \eqref{parabpointgrad} by the definition of $L$ and the fact that $\lambda > 1$. 

There remains to justify \eqref{parabpointgrad} for $|x| > k_3 t$. Recall that we still have the freedom to choose $k_3$ as convenient. By \eqref{parabpoint} and Proposition~\ref{p:estimqetoile}, we have
\begin{equation}
\label{eqeq}
q^\o(t,0,x) \le q^*(kt,x) \le \frac{c}{(1\vee t)^{d/2}} \exp\Ll(- \mcl{D}_{k_2kt}(x)\Rr).
\end{equation}
We now claim that choosing $k_3$ sufficiently large, we can ensure that for every $\td{k} \ge 2 k k_2$,
\begin{equation}
\label{ineqmclD}
|x| \ge k_3 t \ \Rightarrow \ \mcl{D}_{k_2kt}(x) \ge t + \mcl{D}_{\td{k} t}(x).
\end{equation}

Let us first see why \eqref{ineqmclD} enables to conclude. It follows from \eqref{eqeq} and \eqref{ineqmclD} that
$$
q^\o(t,0,x) \le e^{-t} \frac{c}{(1\vee t)^{d/2}} \exp\Ll(- \mcl{D}_{\td{k}t}(x)\Rr).
$$
By Proposition~\ref{p:estimqetoile}, this is smaller than
$$
c e^{-t} q^*(\td{k} t / k_1, x),
$$
for every $\td{k} \ge 2 k k_2$ and every $x, t$ satisfying $|x| \ge k_3 t$. This clearly implies \eqref{parabpointgrad} for $|x| \ge k_3 t$. The proof is thus complete, provided we show \eqref{ineqmclD}.


In order to do so, we first note that
$$
t \Ll( \sqrt{1+\frac{|x|^2}{t^2}} - 1 \Rr) = \sqrt{t^2  + |x|^2} - t = \int_0^{|x|} \frac{u}{\sqrt{t^2 + u^2}} \ \d u \le |x|.
$$
The right-hand side of \eqref{ineqmclD} thus holds as soon as 
\begin{equation}
\label{ineq22}
|x| \arsinh\Ll( \frac{|x|}{k_2 k t} \Rr) \ge t + |x| + |x|\arsinh\Ll( \frac{|x|}{2 k_2 k t} \Rr).
\end{equation}
If $k_3$ is sufficiently large (and in particular $\ge 1$), then it is clear that $|x| \ge k_3 t$ implies
$$
\arsinh\Ll( \frac{|x|}{k_2 k t} \Rr) \ge 2 + \arsinh\Ll( \frac{|x|}{2 k_2 k t} \Rr),
$$
which in turn implies \eqref{ineq22}.
\end{proof}
%
%
The bound \eqref{parabpointgrad} on the gradient of the heat kernel cannot be improved in general (see for instance \cite[p.\ 364]{deldeu}). However, upper bounds matching the homogeneous case can be recovered if one takes averages over the randomness of the stationary field of conductances. For simplicity, we write $q\p(t,x,y)$ instead of $q^{\o\p}(t,x,y)$. The following result is due to \cite[Theorem~1.4]{connadny} and \cite[Theorem~1.1]{deldeu}.
%
%
\begin{prop}[averaged control of the heat kernel \cite{connadny,deldeu}]
There exist constants $c, k > 0$ such that for every $p \in [0,1]$, $t \ge 0$, $x \in \Z^d$ and $i \neq j \in \{2,3\}$, 
\begin{equation}
\label{parabaverage1}
\Ll( \E[|\na_i q\p(t,0,x)|^2]  \Rr)^{1/2} \le c \ \frac{q^*(k t, x)}{\sqrt{1 \vee t}},
\end{equation}
\begin{equation}
\label{parabaverage2}
 \E[|\na_i \na_j q\p(t,0,x)|]   \le c \ \frac{q^*(k t, x)}{{1 \vee t}}.
\end{equation}
\end{prop}
%
For every $\omega \in \Omega$, $\mu > 0$ and $x,y \in \Z^d$, we define the Green function 
$$
G_\mu^\o(x,y) = \int_0^{+\infty} e^{-\mu t} q^\o(t,x,y) \ \d t.
$$
Note that the function $(G_\mu^\omega(x,y))_{y \in \Z^d}$ satisfies
$$
\mu G^\o_\mu(x,y) - \na^* \cdot A(y,\omega) \na G^\o_\mu(x,y) = \1_{x = y} \qquad (y \in \Z^d),
$$
where $\na^*$ and $\na$ are understood as acting on the $y$ variable. Note also that the Green function is symmetric, that is, $G_\mu^\o(x,y) = G_\mu^\o(y,x)$. The estimates obtained on the heat kernel transfer into estimates on the Green function by integration.

%(* Say a word on taking $\mu = 0$ ? *)


\begin{prop}[pointwise control of the Green function]
There exist $c, \td{c} > 0, \alpha > 0$ such that for every $\mu \in (0,1/2]$, $\omega \in \Omega$, $x \in \Z^d$ and $i \in \{1,2\}$,
\begin{equation}
\label{greenpoint1}
|G^\o_\mu(0,x)| \le 
c \left|
\begin{array}{ll}
\displaystyle{
\log(\mu^{-1}) \ e^{-\td{c} \sqrt{\mu} |x|} 
}	& \text{if } d = 2, \\
\displaystyle{
\frac{1}{(1\vee|x|)^{d-2}} \ e^{-\td{c} \sqrt{\mu} |x|}
}	& \text{if } d \ge 3,
\end{array}
\right.
\end{equation}
\begin{equation}
\label{greenpoint2}
|\nabla_i G^\o_\mu(0,x)| \le \frac{c}{(1\vee|x|)^{d-2+\alpha}} \ e^{-\td{c} \sqrt{\mu} |x|}.
\end{equation}
\end{prop}
\begin{proof}
We begin by justifying \eqref{greenpoint1} for $d \ge 3$. In view of \eqref{parabpoint}, it suffices to bound
\begin{equation}
\label{ineq1}
\int_0^{+\infty} e^{-\mu t} p^*(t,x) \ \d t = \int_0^{|x|} e^{-\mu t} p^*(t,x) \ \d t + \int_{|x|}^{+\infty} e^{-\mu t} p^*(t,x) \ \d t.
\end{equation}
Using \eqref{gaussbounds}, we can bound the second integral in the right-hand side above, up to a constant, by
\begin{equation}
\label{integr1}
\int_0^{+\infty} \frac{e^{-\mu t}}{(1 \vee t)^{d/2}} e^{-|x|^2/c_2 t} \ \d t,
\end{equation}
which, by a change of variables, can be bounded by
$$
|x|^{-(d-2)} \int_0^{+\infty} \frac{e^{-\mu|x|^2 u}}{u^{d/2}} e^{-1/c_2  u} \ \d u.
$$
Moreover,
\begin{eqnarray*}
&& \int_0^{+\infty} \frac{e^{-\mu|x|^2 u}}{u^{d/2}} e^{-1/c_2 u} \ \d u \\
& \le & e^{-\sqrt{\mu} |x| /2c_2}  \int_0^{(\sqrt{\mu} |x|)^{-1}} \frac{e^{-1/2 c_2 u}}{u^{d/2}} \ \d u + e^{-\sqrt{\mu} |x|} \int_{(\sqrt{\mu} |x|)^{-1}}^{+\infty} \frac{e^{-1/c_2 u}}{u^{d/2}} \ \d u \\
& \le & C (e^{-\sqrt{\mu} |x| /2c_2} + e^{-\sqrt{\mu}|x|}).
\end{eqnarray*}
We have thus obtained that for $d \ge 3$, there exists $c,\td{c} > 0$ such that
$$
\int_{|x|}^{+\infty} e^{-\mu t} p^*(t,x) \ \d t \le \frac{c}{|x|^{d-2}} e^{-\td{c} \sqrt{\mu} |x|}.
$$
The first integral in the right-hand side of \eqref{ineq1} can be bounded more easily. Indeed, it follows from Proposition~\ref{p:estimqetoile} that for $t < |x|$, we have $p^*(t,x) \le e^{-\td{c} |x|}$ for some $\td{c} > 0$, and thus
$$
\int_0^{|x|} e^{-\mu t} p^*(t,x) \ \d t \le |x| e^{-\td{c}|x|}.
$$
To summarize, we have thus shown that
$$
\int_0^{+\infty} e^{-\mu t} p^*(t,x) \ \d t \le \frac{c}{|x|^{d-2}} e^{-\td{c} \sqrt{\mu} |x|}.
$$
In order to conclude, it then suffices to see that the Green function is bounded uniformly over $\mu$. But this is clear since $\int p^*(t,0)  \d t$ is finite.

When $d = 2$, one needs to be more careful when considering the integral appearing in \eqref{integr1}. It is straightforward to check that this integral is bounded by a constant times $\log(\mu^{-1})$, uniformly over $x$. If $\sqrt{\mu}|x| \le 1$, then there is nothing more to prove. Else, we split the integral along
$$
\Ll(\int_0^1 + \int_1^{|x|/\sqrt{\mu}} + \int_{|x|/\sqrt{\mu}}^{+\infty} \Rr) \frac{e^{-\mu t}}{1 \vee t} \ e^{-|x|^2/c_2 t} \ \d t.
$$
The first integral is bounded by $e^{-|x|^2/c_2}$, and can thus be discarded. For the second one, we obtain the bound
$$
e^{-\sqrt{\mu}|x|/c_2} \int_1^{+\infty}\frac{e^{-\mu t}}{t} \ \d t \le 2 \log(\mu^{-1}) e^{-\sqrt{\mu}|x|/c_2},
$$
while the third integral can be bounded by
$$
e^{-\sqrt{\mu}|x|/2} \int_{|x|/\sqrt{\mu}}^{+\infty} \frac{e^{-\mu t /2}}{t} \ \d t \le \frac{e^{-\sqrt{\mu}|x|}}{\sqrt{\mu}|x|} ,
$$
and the claim is thus proved. 

Inequality \eqref{greenpoint2} is obtained from \eqref{parabpointgrad}, following the same reasoning as for the case $d \ge 3$ of the proof of \eqref{greenpoint1}.
\end{proof}

We write $G\p$ for $G^{\o\p}$.

\begin{prop}[averaged control of the Green function]
\label{p:greenaverage}
There exists $c, \td{c} > 0$ such that for every $p \in [0,1]$, $\mu > 0$, $x \in \Z^d$ and $i \neq j \in \{1,2\}$,
\begin{equation}
\label{greenaverage1}
\Ll( \E[|\nabla_i G\p_\mu(0,x)|^2] \Rr)^{1/2} \le \frac{c}{(1\vee|x|)^{d-1}} \ e^{-\td{c} \sqrt{\mu} |x|},
\end{equation}
\begin{equation}
\label{greenaverage2}
\E[|\nabla_i\nabla_j G\p_\mu(0,x)|]  \le \frac{c}{(1\vee|x|)^{d}} \ e^{-\td{c} \sqrt{\mu} |x|},
\end{equation}
\end{prop}
\begin{proof}
Recall that Minkowski's integral inequality ensures that for any positive measurable function $f$, 
$$
\E\Ll[ \Ll( \int f(s) \ \d s \Rr)^2\Rr] %& = & \int \E[f(s) f(t)] \ \d s \ \d t \\
%& \le & \int \E[f(s)^2]^{1/2} \ \E[f(t)^2]^{1/2}\ \d s \  \d t \\
%& \le & 
\le \Ll( \int \E[f(s)^2]^{1/2} \ \d s  \Rr)^2.
$$
Applying this observation to 
$$
\E[|\nabla_i G\p_\mu(0,x)|^2] \le \E\Ll[ \Ll(\int e^{-\mu t} |\na_i q\p(t,0,x)| \ \d t\Rr)^2 \Rr],
$$
we get that
$$
\Ll( \E[|\nabla_i G\p_\mu(0,x)|^2] \Rr)^{1/2} \le \int_0^{+\infty} e^{-\mu t} \ \E\Ll[|\na_i q\p(t,0,x)|^2\Rr]^{1/2} \ \d t.
$$
The rest of the proof of \eqref{greenaverage1} is the same as that of \eqref{greenpoint1}, proceeding by integration of the estimate obtained in \eqref{parabaverage1}. The proof of \eqref{greenaverage2} is identical, integrating estimate \eqref{parabaverage2}.
\end{proof}

%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Solving regularized elliptic equations}
\label{s:solve}
%\setcounter{equation}{0}

We say that a function $f : \Z^d \to \R$ is of \emph{sub-exponential growth} if for every $\delta > 0$, we have $|f(x)| = O(e^{\delta|x|})$ as $|x|$ tends to infinity.

\begin{thm}[Existence and uniqueness of solutions]
\label{t:ex-un}
Let $\mu > 0$, and $f: \Z^d \to \R$ be of sub-exponential growth. For every $\omega \in \Omega$, there exists a unique function $\chi : \Z^d \to \R$ of sub-exponential growth satisfying
\begin{equation}
\label{eqphichi}
\mu \chi - \na^* \cdot A^\o \na \chi = f  \qquad (\text{in } \Z^d).
\end{equation}
Moreover, $\chi$ is given by
\begin{equation}
\label{defphichi}
\chi(x) = \sum_{y \in \Z^d} G^\o_\mu(x,y) \ f(y) \qquad (x \in \Z^d).
\end{equation}
\end{thm}
\begin{proof}
Since \eqref{greenpoint1} ensures exponential decay of the Green function, it is easy to see that the function $\chi$ written in \eqref{defphichi} is well-defined, of sub-exponential growth, and satisfies equation \eqref{eqphichi}. There remains to show uniqueness, and for this, it suffices to consider the case $f = 0$. Let $B_n = \{-n,\ldots,n\}^d$ be the box of size $n$. We have
\begin{equation}
\label{e:unique}
\sum_{x \in B_n} \chi(x)\Ll(\mu \chi(x) - \na^* \cdot A^\o(x) \na \chi(x) \Rr) = 0.
\end{equation}
We want to perform an integration by parts. The difference between 
$$
- \sum_{x \in B_n} \chi(x) \na^* \cdot A^\o(x) \na \chi(x)
$$
and 
\begin{equation}
\label{e:unique2}
\sum_{x \in B_n} \na \chi(x) \cdot A^\o(x) \na \chi(x)
\end{equation}
is bounded by
$$
\sum_{\substack{x \in B_n, y \in B_{n+1} \\ x \sim y}} |\chi(x)| \ |A^\o(y) \na \chi(y)| + |\chi(y)| \ |A^\o(x) \na \chi(x)|,
$$
which, up to a constant, is bounded by
$$
\sum_{x \in B_{n+2} \setminus B_{n-2}} \chi(x)^2.
$$
Hence, combining this estimate with \eqref{e:unique} and the fact that the term in \eqref{e:unique2} is positive, we obtain
$$
\mu \sum_{x \in B_n} \chi(x)^2 \le c \sum_{x \in B_{n+2} \setminus B_{n-2}} \chi(x)^2,
$$
for some constant $c > 0$. In a more condensed form, if we write 
$$
u_n =\sum_{x \in B_n} \chi(x)^2,
$$
we have thus obtained that $\mu \ u_n \le c (u_{n+2} - u_{n-2})$. Since $u_n$ is increasing, we have shown that 
$$
u_{n+2} \ge \Ll(1+\frac{\mu}{c}\Rr)u_{n-2}.
$$
If $u_n$ is not identically zero, then it must grow exponentially fast. But this would contradict the assumption that $\chi$ is of sub-exponential growth. Hence $u_n$ is identically zero, and so is $\chi$.
\end{proof}

%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Approximating the homogenized matrix}
\label{s:loc}
%\setcounter{equation}{0}
Let $\mu > 0$. For every $p \in [0,1]$ and $\un{\o} \in \un{\O}$, we let $\phi_\mu\p(\cdot,\un{\o}) : \Z^d \to \R$ be the unique function of sub-exponential growth satisfying
\begin{equation}
\label{defphimu}
\mu \phi_\mu\p - \na^* \cdot A\p (\xi + \na \phi_\mu\p) = 0.
\end{equation}
We write $\phi_\mu\z$ for $\phi_\mu^{(0)}$. The introduction of the function $\phi_\mu\p$ is interesting for our purpose as an approximation of $\phi\p$. The former is more localized than $\phi\p$, in the sense that, roughly speaking, the value of $\phi_\mu\p$ at a point depends only on the values of the conductances in a box of size $\mu^{-1/2}$.

A standard energy estimate shows that $\E[|\na \phi_\mu\p|^2]$ is bounded uniformly over $\mu$ and $p$ (we recall that whenever we write an expression like $\E[ |\na \phi_\mu\p|^2 ]$, we understand that the function under the expectation is evaluated at $(0,\un{\o})$). We will need the following much more precise information on the corrector, which is borrowed from \cite[Proposition~2.1]{glotto}.
%
\begin{thm}[finite moments of the corrector, \cite{glotto}]
\label{t:corr-moments}
For every $k \in \N$, there exist constants $c$ and $r \ge 0$ such that for every $\mu \in (0,1/2]$ and $p \in [0,1]$, 
$$
\E\Ll[ | \phi_\mu\p |^k \Rr] \le 
c \left| 
\begin{array}{ll}
	 \log^r(\mu^{-1}) & \text{if } d = 2, \\
	1  & \text{if } d \ge 3.
\end{array}
\right.
$$
\end{thm}

It is crucial for our subsequent arguments to have a good quantitative estimate on the difference between the homogenized matrix and the approximation obtained through $\phi_\mu\p$. The following result is what we need, and follows from \cite[Theorem~1]{glotto2} (or also from \cite{glotto} and \cite[Proposition~9.1 with $k = 1$]{vardecay}).

\begin{thm}[approximation of $\Ah$ based on $\phi_\mu$, \cite{glotto,vardecay, glotto2}] 
There exist constants $c$ and $r \ge 0$ such that for every $\mu \in (0,1/2]$ and $p \in [0,1]$,
$$
\Ll|\xi \cdot \Ah\p \xi - \E[\xi \cdot A\p(\xi + \na \phi_\mu\p)] \Rr| \le c
\left|
\begin{array}{ll}
	\mu \log^r(\mu^{-1}) & \text{if } d = 2, \\
	\mu  & \text{if } d \ge 3.
\end{array}
\right.
$$
\label{t:approx}
\end{thm}

%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Linear approximation of the corrector}
\label{s:linear}
%\setcounter{equation}{0}



Each edge $e$ can be written uniquely as $(z,z+\e_i)$ for some $z \in \Z^d$ and $1 \le i \le d$. We write $\un{e} = z$. %For $f: \Z^d \to \R$, we may write $f(e)$ to denote $f(z)$. 
We define $C^e : \Z^d \times \un{\Omega} \to \mcl{M}_d$ (the set of $d \times d$ matrices) by $C^e(x,\un{\o}) = 0$ (the matrix whose elements are all $0$) if $x \neq \un{e}$, and $C^e(\un{e},\un{\o})$ to be the diagonal matrix whose diagonal elements are all $0$ except the $i^\text{th}$ one, which is equal to ${\o}^{(1)}_e - \o_e$. As usual, we may keep the variable $\un{\o}$ implicit if clear from the context. 
We let $A^e = A^\circ + C^e$. In words, the environment corresponding to $A^e$ is perturbed only at the edge $e$. More generally, for every $E \subset \B$, we define
\begin{equation}
\label{defAE}
C^E = \sum_{e \in E} C^e \quad \text{ and } \quad A^E = A^\circ + C^E,
\end{equation}
%
$G_\mu^E$ for the Green function corresponding to the environment given by $A^E$, and $E\p = \{e \in \B : b\p_e = 1\}$. Note that by definition, $A\p = A^{E\p}$ and $G_\mu\p = G_\mu^{E\p}$. For every $E \subset \B$, we let $\phi^E_\mu : \Z^d \times \un{\Omega} \to \R$ be the unique function of sub-exponential growth such that
\begin{equation}
\label{defphiE}
\mu \phi^E_\mu - \na^* \cdot A^E (\xi + \na \phi^E_\mu) = 0,
\end{equation}
as given by Theorem~\ref{t:ex-un}, so that
\begin{equation}
\label{defphiE-green}
\phi^E_\mu(x) = \sum_{y \in \Z^d} G^E_\mu(x,y) \ \na^* \cdot A^E(y) \xi = - \sum_{y \in \Z^d} \na_2 G^E_\mu(x,y)  \cdot A^E(y) \xi,
\end{equation}
where the integration by part in the last equality is justified since $G$ decays exponentially fast, see \eqref{greenpoint1}. We have $\phi\p = \phi^{E\p}$.
We write $\phi^e_\mu$ as a shorthand for $\phi^{\{e\}}_\mu$. 
We also define
\begin{equation}
\label{deftdphi}
\td{\phi}^E_\mu = \phi\z_\mu + \sum_{e \in E} (\phi^e_\mu - \phi\z_\mu)%\ov{\phi}^e_\mu
,
\end{equation}
(we will see shortly that this definition makes sense), and write $\td{\phi}\p_\mu$ for $\td{\phi}^{E\p}_\mu$. The purpose of this section is to prove the following result, which roughly speaking, states that in the limit of diluted perturbations, the perturbed corrector is close to its linear approximation.

\begin{thm}[linear approximation of the corrector]
\label{t:approxcorr}
There exist constants $c$, $\beta > 0$, $\gamma > 0$ such that for every $\mu \in (0,1/2]$ and $p \in [0,1]$,
$$
\E\Ll[\Ll|\na \phi\p_\mu - \na \td{\phi}\p_\mu \Rr|\Rr] \le c p^2 \mu^{-1+\beta} + e^{-\mu^{-\gamma}}.
$$
\end{thm}
%
\begin{rem}
We will see in section~\ref{s:error} that the upper bound can be improved using the recent results of \cite{marott}, see Theorem~\ref{t:sharp}.
\end{rem}
We start with several lemmas concerning the effect of perturbations on the corrector or the Green function.
%
\begin{lem}[perturbation at one edge]
\label{l:one}
Let
\begin{equation}
\label{defovphie}
\ov{\phi}^e_\mu = \phi^e_\mu - \phi\z_\mu.
\end{equation}
For every $\mu > 0$ and $x \in \Z^d$, we have
$$
\ov{\phi}^e_\mu(x) =- \na_2 G^e_\mu(x,\un{e}) \cdot C^e(\un{e}) (\xi + \na \phi\z_\mu(\un{e})).
$$
\end{lem}
\begin{proof}
In view of equation \eqref{defphimu} satisfied by $\phi_\mu\z$ and of the identity $A^e = A\z + C^e$, we infer that
$$
\mu \phi\z_\mu - \na^* \cdot A^e (\xi + \na \phi\z_\mu) = - \na^* \cdot C^e (\xi + \na \phi\z_\mu).
$$
This and equation \eqref{defphiE} satisfied by $\phi_\mu^e$ lead to
\begin{equation}
\label{propovphie}
\mu \ov{\phi}^e_\mu - \na^* \cdot A^e \na \ov{\phi}^e_\mu = \na^* \cdot C^e (\xi + \na \phi\z_\mu).
\end{equation}
The function on the right-hand side above is non-zero only at a finite number of points, so it is clearly of sub-exponential growth. The function $\ov{\phi}^e_\mu$ is also of sub-exponential growth by \eqref{defphiE-green} and \eqref{greenpoint1}. By Theorem~\ref{t:ex-un}, 
\begin{eqnarray*}
\ov{\phi}^e_\mu(x) & = & \sum_y G^e_\mu(x,y) \ \na^*\cdot C^e (\xi + \na \phi\z_\mu)(y) \\
& = & -\sum_y \na_2 G^e_\mu(x,y) \cdot C^e(y) (\xi + \na \phi\z_\mu(y)) \\
& = & -\na_2 G^e_\mu(x,\un{e}) \cdot C^e(\un{e}) (\xi + \na \phi\z_\mu(\un{e})),
\end{eqnarray*}
where in the second equality, the integration by parts is justified since only a finite number of terms in the sum are non-zero.
\end{proof}
\begin{rem}
\label{r:deftdphi}
As was recalled in the beginning of section~\ref{s:loc}, the gradient of $\phi_\mu\z$ is square-integrable: $\E[|\na \phi\z_\mu|^2] < \infty$. hence, by the ergodic theorem, 
\begin{equation}
\label{e:ergodic}
\frac{1}{|B_n|} \sum_{x \in B_n} |\na \phi_\mu\z(x)| \xrightarrow[n \to \infty]{\text{a.s.}} \E[|\na \phi_\mu\z|] < +\infty.
\end{equation}
It thus follows from Lemma~\ref{l:one} and the decay of the Green function \eqref{greenpoint1} that for every $x$, $\ov{\phi}_\mu^e(x)$ decays exponentially fast with the distance from $x$ to the edge $e$. In particular, for every $E \subset \B$, the function~$\td{\phi}_\mu^E$ defined in \eqref{deftdphi} is always well-defined and of sub-exponential growth.
\end{rem}

\begin{lem}[Perturbation at several edges]
\label{l:several}
Let $E \subset \B$, and let
\begin{equation}
\label{defovphip}
\ov{\phi}^E_\mu = \phi^E_\mu - \td{\phi}^E_\mu.
\end{equation}
For every $\mu > 0$ and $x \in \Z^d$, we have
$$
\ov{\phi}^E_\mu(x) = - \sum_{e \neq e' \in E} \na_2 G^E_\mu(x,\un{e}') \cdot C^{E\setminus \{e \}}(\un{e}') \na \ov{\phi}^e_\mu(\un{e}').
$$
\end{lem}
\begin{proof}
If $e \in E$, then using \eqref{propovphie} and the fact that $A^E = A^e + C^{E\setminus\{e\}}$, we have
$$
\mu \ov{\phi}^e_\mu - \na^* \cdot A^E \na \ov{\phi}^e_\mu = \na^* \cdot C^e (\xi + \na \phi\z_\mu) - \na^* \cdot C^{E\setminus \{e \}} \na \ov{\phi}^e_\mu.
$$
Similarly, we have
$$
\mu \phi\z_\mu - \na^* \cdot A^E (\xi + \na \phi\z_\mu) = - \na^* \cdot C^E (\xi + \na \phi\z_\mu).
$$
Recalling that
$$
\ov{\phi}^E_\mu = \phi^E_\mu - \phi\z_\mu - \sum_{e \in E} \ov{\phi}^e_\mu,
$$
that $C^E = \sum_{e \in E} C^e$, and equation \eqref{defphiE} satisfied by $\phi^E_\mu$, we obtain
$$
\mu \ov{\phi}^E_\mu - \na^* \cdot A^E \na \ov{\phi}^E_\mu = \sum_{e \in E} \na^* \cdot C^{E\setminus \{e \}} \na \ov{\phi}^e_\mu.
$$
By Remark~\ref{r:deftdphi}, the function $\ov{\phi}^E_\mu$ is of sub-exponential growth. Moreover, arguing as in this remark, we see that the function in the right-hand side above is also of sub-exponential growth. By Theorem~\ref{t:ex-un}, we thus have
\begin{eqnarray*}
\ov{\phi}^E_\mu(x) & = & \sum_{e \in E} \sum_{y \in \Z^d}  G^E_\mu(x,y) \ \na^* \cdot C^{E\setminus \{e \}} \na \ov{\phi}^e_\mu(y) \\
& = & - \sum_{e \in E} \sum_{y \in \Z^d} \na_2 G^E_\mu(x,y) \cdot C^{E\setminus \{e \}}(y) \na \ov{\phi}^e_\mu(y) \\
& = & - \sum_{e \neq e' \in E} \na_2 G^E_\mu(x,\un{e}') \cdot C^{E\setminus \{e \}}(\un{e}') \na \ov{\phi}^e_\mu(\un{e}'),
\end{eqnarray*}
where the integration by parts in the second equality is justified using the exponential decay of the Green function and the fact that $y \mapsto |\ov{\phi}^e_\mu(y)|$ is of sub-exponential growth, see Lemma~\ref{l:one} and \eqref{e:ergodic}.
\end{proof}
%
%
%
%
%
%
\begin{lem}[averaged control of the perturbed Green function]
For every $\zeta \in (0,1]$, there exists $c > 0$ such that for every $\mu \in (0,1/2]$ and $e,e' \in \B$,
$$
\E\Ll[ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})|^{1/\zeta}  \Rr] \le \frac{c}{(1\vee |\un{e}-\un{e}'|)^d}.
$$
\label{l:greenperturbed}
\end{lem}
\begin{proof}%[Proof of Lemma~\ref{l:greenperturbed}] 
As a first step, we show that the lemma is true if the perturbed Green function $G^e_\mu$ is replaced by the non-perturbed one $G\z_\mu$. %For convenience, we write $1/\zeta$ as $1+\delta$ for some $\delta \in (0,1)$. 

We know from \eqref{greenaverage2} that
$$
\E\Ll[ | \na_1 \na_2 G\z_\mu(\un{e}',\un{e})| \Rr] \le \frac{c}{(1\vee |\un{e}-\un{e}'|)^d}.
$$
Moreover, by \eqref{greenpoint2}, the quantity $\na_2G^\o_\mu(x,y)$ is bounded uniformly over $\o \in \O$, $x,y \in \Z^d$ and $\mu > 0$. In particular, the random variable 
$$
| \na_1 \na_2 G\z_\mu(\un{e}',\un{e})|
$$
is bounded. Hence, the lemma is true if $G^e_\mu$ is replaced by $G\z_\mu$.



%For any positive random variable $X$, H\"older's inequality gives
%$$
%\E[X^{1+\delta}] = \E[X^{1-\delta + 2\delta}] \le \E[X]^{1-\delta} \ \E[X^2]^{\delta}.
%$$
%We use this inequality with $X = | \na_1 \na_2 G\z_\mu(\un{e}',\un{e})|$. By \eqref{greenaverage2}, we have
%$$
%\E[X] \le \frac{c}{(1\vee |\un{e}-\un{e}'|)^d}.
%$$
%On the other hand, inequality \eqref{greenaverage1} ensures that 
%$$
%\E[X^2]^{1/2} \le \frac{c}{(1\vee |\un{e}-\un{e}'|)^{d-1}}.
%$$
%Combining these two results, we obtain
%$$
%\E[X^{1+\delta}] \le \frac{c}{(1\vee |\un{e}-\un{e}'|)^{d(1-\delta)+2\delta(d-1)}},
%$$
%and we have indeed $d(1-\delta)+2\delta(d-1) \ge d$ (recall that we assume $d \ge 2$).

Let us write $\ov{G}^e_\mu = G^e_\mu - G\z_\mu$. In order to prove the lemma, it now suffices to argue that 
$$
\E\Ll[ | \na_1 \na_2 \ov{G}^e_\mu(\un{e}',\un{e})|^{1/\zeta}  \Rr]\le \frac{c}{(1\vee |\un{e}-\un{e}'|)^d}.
$$
Let $x \in \Z^d$. The Green function $G\z_\mu(x,\cdot)$ satisfies
$$
\mu G\z_\mu(x,\cdot) -\na^* \cdot A\z \na G\z_\mu(x,\cdot) = \1_x.
$$
The same equation holds for $G^e_\mu(x,\cdot)$ provided $A\z$ is replaced by $A^e = A\z + C^e$, hence
$$
\mu G^e_\mu(x,\cdot) -\na^* \cdot A\z \na G^e_\mu(x,\cdot) = \1_x + \na^* \cdot C^e \na G^e_\mu(x,\cdot).
$$
Combining the two previous identities, we obtain that
$$
\mu \ov{G}^e_\mu(x,\cdot) - \na^* \cdot A\z \na \ov{G}^e_\mu(x,\cdot) = \na^* \cdot C^e \na G^e_\mu(x,\cdot).
$$
By Theorem~\ref{t:ex-un}, we are led to
\begin{eqnarray*}
\ov{G}^e_\mu(x,y) & = & \sum_{z \in \Z^d} G\z_\mu(y,z) \ \na^* \cdot C^e \na G^e_\mu(x,\cdot)(z) \\
& = & - \sum_{z \in \Z^d} \na_2 G\z_\mu(y,z) \ \cdot C^e \na G^e_\mu(x,\cdot)(z) \\
& = & \na_2G\z_\mu(y,\un{e}) \ \cdot C^e(\un{e}) \na_2 G^e_\mu(x,\un{e}),
\end{eqnarray*}
and by symmetry of the Green functions, we also have 
\begin{equation}
\label{eqMO}
\ov{G}^e_\mu(x,y) = \na_2G\z_\mu(x,\un{e}) \ \cdot C^e(\un{e}) \na_2 G^e_\mu(y,\un{e}).
\end{equation}
It thus follows that
\begin{equation}
\label{relovG}
\na_1 \na_2\ov{G}^e_\mu(x,y) = C^e(\un{e}) \Ll(\na_1 \na_2 G^e_\mu(y,\un{e})\Rr) \ \Ll(\na_1 \na_2G\z_\mu(x,\un{e})\Rr),
\end{equation}
which we need to evaluate for $x = \un{e}'$ and $y = \un{e}$. Since, by \eqref{greenpoint2},
$$
\Ll|  C^e(\un{e}) \na_1 \na_2 G^e_\mu(\un{e},\un{e}) \Rr|
$$
is bounded uniformly over $\mu$ and $\un{\o}$, it suffices to see that
$$
\E\Ll[ |\na_1 \na_2G\z_\mu(\un{e}',\un{e})|^{1/\zeta} \Rr] \le \frac{c}{(1\vee |\un{e}-\un{e}'|)^d},
$$
but this was obtained during the first step of this proof, so we are done.
\end{proof}
%
%
%
%
\begin{proof}[Proof of Theorem~\ref{t:approxcorr}]
Note first that
$$
\E\Ll[\Ll|\na \phi\p_\mu - \na \td{\phi}\p_\mu \Rr|\Rr] = \E\Ll[\Ll|\na \ov{\phi}\p_\mu\Rr|\Rr],
$$
where as usual, we write $\ov{\phi}\p_\mu = \ov{\phi}^{E\p}_\mu$. By Lemma~\ref{l:several}, we have
\begin{equation*}
\na \ov{\phi}\p_\mu(0) = - \sum_{e \neq e' \in E\p} \Ll(\na_1 \na_2 G\p_\mu(0,\un{e}') \Rr) \ C^{E\p\setminus \{e \}}(\un{e}') \na \ov{\phi}^e_\mu(\un{e}'),
\end{equation*}
so we have the bound
\begin{equation*}
%\label{e:gradovphip}
|\na \ov{\phi}\p_\mu(0)| \le c_+ \sum_{e \neq e' \in E\p} |\na_1 \na_2 G\p_\mu(0,\un{e}')| \ |\na \ov{\phi}^e_\mu(\un{e}')|.
\end{equation*}
By Lemma~\ref{l:one}, the right-hand side above is bounded, up to a constant, by
\begin{equation}
\label{e:gradovphip}
\sum_{e \neq e' \in E\p}| \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})|.
\end{equation}

Let $\gamma > 1/2$, and write $E\p_\mu = E\p \cap B_{\mu^{-\gamma}}$, where we recall that $B_n = \{-n, \ldots, n\}^d$. We will argue that in the above sum, the only terms which truly contribute to the sum are those for which $e,e' \in E\p_\mu$. In order to do so, it is convenient to introduce $\td{\gamma}$ such that $1/2 < \td{\gamma} < \gamma$, and to define $\td{E}\p_\mu = E\p \cap B_{\mu^{-\td{\gamma}}}$.

Recall that $\E[|\na \phi\z_\mu(\un{e})|] \le \E[|\na \phi\z_\mu(\un{e})|^2]^{1/2}$ does not depend on $e$ and is bounded uniformly over $\mu > 0$. We also have from \eqref{greenpoint2} that for every $E \subset \B$,
\begin{equation}
\label{greepointsimpl}
|\na_2 G^E_\mu(x,y)| \le c \ e^{-\td{c} \sqrt{\mu} |y-x|},
\end{equation}
with $\td{c} > 0$. Hence, up to a constant, the expectation
\begin{equation}
\label{e:gradovphip2}
\E\Ll[ \sum_{e' \in E\p \setminus \td{E}\p_\mu} \ \sum_{e \in E\p, e \neq e'}  | \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})| \Rr]
\end{equation}
is bounded by
$$
\sum_{z' \in \Z^d \setminus B_{\mu^{-\td{\gamma}}}} \ \sum_{z \in \Z^d} e^{-\td{c}\sqrt{\mu} |z'|} \ e^{-\td{c}\sqrt{\mu} |z-z'|}.
$$
Summation over $z$ gives a contribution of order $\mu^{-d/2}$, while the remaining summation over $z'$ decays as $e^{-\mu^{-\gamma'}}$ for every $\gamma' < \td{\gamma}-1/2$. We have thus justified that the difference between the expectation of \eqref{e:gradovphip} and \eqref{e:gradovphip2} is $O(e^{-\mu^{-\gamma'}})$ for every $\gamma' < \td{\gamma}-1/2$. We can thus focus on studying
$$
\E\Ll[ \sum_{e' \in  \td{E}\p_\mu, e \in E\p, e \neq e'}  | \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})| \Rr].
$$
A similar reasoning enables to show that in the above sum, the terms for which $e \notin E\p_\mu$ can be discarded. Indeed, the contribution of those terms can be bounded, up to a constant, by
\begin{equation}
\label{eq1234}
\sum_{z' \in B_{\mu^{-\td{\gamma}}}} \ \sum_{z \in \Z^d \setminus B_{\mu^{-\gamma}}} e^{-\td{c} \sqrt{\mu} |z'|} \ e^{-\td{c} \sqrt{\mu} |z-z'|}.
\end{equation}
Since $\td{\gamma} < \gamma$, the distance between $z$ and $z'$ in any term of this double sum is always greater than $\mu^{-\gamma}/2$. Hence, we obtain an upper bound for the inner sum in \eqref{eq1234} if we sum over all $z$ such that $|z-z'| \ge \mu^{-\gamma}/2$. This gives a sum of order $O(e^{-\mu^{-\gamma'}})$ for some $\gamma' > 0$, and hence a similar bound holds for the total sum in \eqref{eq1234}. We have thus argued that it suffices to study
$$
\E\Ll[ \sum_{e' \in  \td{E}\p_\mu, e \in E\p_\mu, e \neq e'}  | \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})| \Rr].
$$
Finally (in order to keep light notations), arguing as in the first step, we see that we can as well consider the sum as ranging over all $e, e' \in E\p_\mu$, $e \neq e'$. 

We now decompose the expectation into
\begin{multline}
\label{e:thestep}
\E\Ll[ \sum_{e\neq e' \in  E\p_\mu}  | \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})| \Rr] \\
= \sum_E \sum_{e\neq e' \in E} \mcl{E}(e,e',E) \ \P[E\p_\mu = E],
\end{multline}
where in the first sum, $E$ ranges over all possible subsets of $B_{\mu^{-\gamma}}$, and where we write
$$
\mcl{E}(e,e',E) = \E\Ll[ | \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})| \ \Big| \ E\p_\mu = E\Rr].
$$
For any $\zeta\in (0,1)$, we have by H\"older's inequality
\begin{multline}
\label{e:mcle}
\mcl{E}(e,e',E) \le \E\Ll[ \Ll(| \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})|\Rr)^{1/\zeta} \ \Big| \ E\p_\mu = E\Rr]^\zeta \\
\E\Ll[ |\xi + \na \phi\z_\mu(\un{e})|^{1/(1-\zeta)} \ \Big| \ E\p_\mu = E\Rr]^{1-\zeta}.
\end{multline}
Since $|\xi + \na \phi\z_\mu(\un{e})|$ does not depend on $E\p$, the last expectation is simply
$$
\E\Ll[ |\xi + \na \phi\z_\mu(\un{e})|^{1/(1-\zeta)} \Rr]^{1-\zeta},
$$
which is bounded by a power of $\log(\mu^{-1})$ by Theorem~\ref{t:corr-moments}. 

For the first expectation in the right-hand side of \eqref{e:mcle}, we use the pointwise bound \eqref{greenpoint2} on the gradient of the Green function to get
\begin{equation*}
\begin{split}
& \E\Ll[ \Ll(| \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})|\Rr)^{1/\zeta} \ \Big| \ E\p_\mu = E\Rr]^\zeta \\
& \qquad \le \frac{c}{(1\vee |\un{e}'|)^{d-2+\alpha}} \ \E\Ll[ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})|^{1/\zeta} \ \Big| \ E\p_\mu = E\Rr]^\zeta \\
& \qquad = \frac{c}{(1\vee |\un{e}'|)^{d-2+\alpha}} \ \E\Ll[ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})|^{1/\zeta} \Rr]^\zeta,
\end{split}
\end{equation*}
and Lemma~\ref{l:greenperturbed} ensures that the latter is bounded by
$$
\frac{c}{(1\vee |\un{e}'|)^{d-2+\alpha} \ (1\vee |\un{e}-\un{e}'|)^{\zeta d}}.
$$
It thus follows that the right-hand side of \eqref{e:thestep} is bounded by some power of $\log(\mu^{-1})$ times
\begin{multline*}
\E\Ll[ \sum_{e \neq e' \in E\p_\mu}\frac{1}{(1\vee |\un{e}'|)^{d-2+\alpha} \ (1\vee |\un{e}-\un{e}'|)^{\zeta d}} \Rr] \\
= p^2 \sum_{e \neq e' \in B_{\mu^{-\gamma}} }\frac{1}{(1\vee |\un{e}'|)^{d-2+\alpha} \ (1\vee |\un{e}-\un{e}'|)^{\zeta d}}.
\end{multline*}
To sum up, we have shown that there exists $\gamma' > 0$ such that for every $\zeta < 1$,
$$
\E\Ll[\Ll|\na \ov{\phi}\p_\mu\Rr|\Rr] \le c  p^2 \log^r(\mu^{-1}) \sum_{e \neq e' \in B_{\mu^{-\gamma}} }\frac{1}{(1\vee |\un{e}'|)^{d-2+\alpha} \ (1\vee |\un{e}-\un{e}'|)^{\zeta d}} + ce^{-\mu^{-\gamma'}},
$$
for some exponent $r \ge 0$.
Up to a multiplicative constant, the sum over $e,e'$ above can be bounded by the integral
\begin{eqnarray*}
\int_{|x|, |y| \le \mu^{-\gamma}} \frac{\d x \ \d y}{|x|^{d-2+\alpha}|y-x|^{\zeta d}} &  \le & \int_{|x| \le \mu^{-\gamma}, |y-x| \le 2\mu^{-\gamma}}  \frac{\d x \ \d y}{|x|^{d-2+\alpha}|y-x|^{\zeta d}} \\
& \le & c \mu^{-\gamma(d-\zeta d +2-\alpha)}.
\end{eqnarray*}
Until now, the parameters $\gamma > 1/2$ and $\zeta < 1$ were arbitrary. We can choose them in such a way that
$$
\gamma(d-\zeta d +2-\alpha) < 1,
$$
and this finishes the proof of Theorem~\ref{t:approxcorr}.
\end{proof}



%
%show that 
%\begin{enumerate}
%\item the sum can be restricted to a sum over edges in a box of size $\mu^{-1/2-\gamma}$, any $\gamma > 0$,
%\item 
%$$
%\E[|\nabla^2 G^e_\mu(e',e)|] \le \frac{1}{|e'-e|^d},
%$$
%and still decay in power $d$ if we take the $\delta$ power instead of the first power, any $\delta > 1$ close to $1$.
%\item Follow computation of page 9-10, gives a decay of order 
%$$
%p^2 \int_{|e|,|e'| \le \mu^{-(1/2+\gamma)}} \frac{\d e \ \d e'}{|e|^{d-2+\alpha}|e'-e|^{\delta d}} \sim p^2 \mu^{(1/2+\gamma)(d-2+\alpha+\delta d - 2d)} \simeq p^2 \mu^{-2/d+\eps}.
%$$
%\end{enumerate}

%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Solving stationary elliptic equations}
\label{s:station}
Recall that we say that a function $\psi : \Z^d \times \un{\Omega} \to \R$ is stationary if $\psi(x,\un{\omega}) = \psi(0,\theta_x \ \un{\omega})$, where $(\theta_x)_{x \in \Z^d}$ denotes the action of $\Z^d$ on $\un{\Omega}$. The stationary function $\psi$ is thus fully characterized by the knowledge of the function $\un{\omega} \to \psi(0,\un{\omega})$, with which it can be identified.

The first purpose of this section is to recall existence and uniqueness results for elliptic equations. Contrary to what was done in section~\ref{s:solve}, the equations considered here do not contain a zero-order regularizing parameter. On the other hand, we will work with the extra assumption of stationarity. Because of the identification mentioned above, we will focus on functions defined on $\un{\Omega}$ (instead of $\Z^d \times \un{\Omega}$). We may keep this identification implicit, writing for instance $A\z$ instead of $A\z(0,\cdot)$. 


%
%
For $f : \un{\Omega} \to \R$, we define the forward gradient $D f : \un{\Omega} \to \R^d$ as  
$$
D f(\un{\o})=\left[  
\begin{array}{c}
f(\theta_{\e_1} \ \un{\o})-f(\un{\o}) \\
\vdots\\
f(\theta_{\e_d} \ \un{\o})-f(\un{\o})
\end{array}
\right],
$$
and for $F = (F_1,\ldots,F_d): \un{\O} \to \R^d$, we write $D^* \cdot F$ for the backward divergence, 
$$
D^* \cdot F(\un{\o}) = \sum_{i = 1}^d \Ll(F_i(\un{\o}) - F_i(\theta_{-\e_i} \ \un{\o})\Rr).
$$
We write $\|F\|_2$ for the $L^2$ norm of $F$, that is,
$\|F\|_2^2 = \sum_{i = 1}^d \E[F_i^2]$. We let $L^2(\un{\O})$ be the space of functions $F : \un{\O} \to \R^d$ such that $\|F\|_2$ is finite, and $L^2_\na$ be the closure in $L^2(\un{\O})$ of $\{D f, \ f : \un{\O} \to \R \text{ s.t. } \E[f^2] < \infty \}$. 
\begin{thm}[Existence and uniqueness of solutions]
\label{t:station}
	For every $F \in L^2(\un{\O})$, there exists a unique $\chi \in L^2_\na$ such that% for every square-integrable $f : \un{\O} \to \R$,
\begin{equation}
\label{e:station}
- D^* \cdot A\z \chi =  D^* \cdot F.
\end{equation}
\end{thm}
\begin{proof}
We begin by showing the existence of solutions, following roughly the arguments of \cite{kipvar}. (Although the proof is by now standard, it is useful to recall its workings since we will need to extend it slightly later on.) The operator $\L := -D^* \cdot A\z D$ is self-adjoint and positive on $\{f : \un{\O} \to \R : \E[f^2] < \infty\}$. The positivity comes from the observation that, for every square-integrable $f$,
\begin{equation}
\label{posit}
\E[f \ \L f] = \E[Df \cdot A\z Df]% = \frac{1}{2} \sum_{z \sim 0} \E[\omega_{0,z} (f(\theta_z\ \o) - f(\o))^2]
.
\end{equation}
Hence, for every $\mu > 0$, there exists a square-integrable $\Psi_\mu : \un{\O} \to \R$ such that
\begin{equation}
\label{e:Psi}
(\mu + \L)\Psi_\mu = D^* \cdot F.
\end{equation} 
Let $f  = D^* \cdot F$. Since $f$ is square-integrable, by the spectral theorem, there exists a measure $e_f$ on $\R_+$ such that for every bounded continuous function $G : \R_+ \to \R$, one has
\begin{equation}
\label{e:spectr}
\E\Ll[ f \ G(\L)(f) \Rr] = \int G(\lambda) \ \d e_f(\lambda).
\end{equation}
Note that for every square-integrable function $g$, one has
$$
\E\Ll[f \ g\Rr] = - \E[F \cdot Dg] \le \|F\|_2 \ \|Dg\|_2,
$$
by the Cauchy-Schwarz inequality. Furthermore, by \eqref{posit},
$$
\|Dg\|_2^2 \le \frac{1}{c_-} \E[g \ \L g],
$$
so that 
$$
\E\Ll[f \ g\Rr] \le \frac{\|F\|_2}{\sqrt{c_-}} \ \E[g \ \L g]^{1/2}.
$$
For every $n \in \N$, let $G_n(\lambda) = \lambda^{-1} \wedge n$. Taking $g = G_n(\L)(f)$ in the inequality above, and using \eqref{e:spectr}, we get
$$
\int (\lambda^{-1} \wedge n) \ \d e_f(\lambda) \le \frac{\|F\|_2}{\sqrt{c_-}} \Ll(\int \lambda(\lambda^{-1} \wedge n)^2 \ \d e_f(\lambda)\Rr)^{1/2}.
$$
By the monotone convergence theorem, this turns into
\begin{equation}
\label{estimH-1}
\int \frac{1}{\lambda} \ \d e_f(\lambda) \le \frac{\|F\|_2^2}{c_-}.
\end{equation}
Equipped with this inequality, we can show that $(D\Psi_\mu)_{\mu > 0}$ is a Cauchy sequence in $L^2(\un{\O})$. Indeed, note that by \eqref{posit}, we have
\begin{equation}
\label{e:stepcauchy}
\|D\Psi_\mu - D\Psi_\nu\|_2^2 \le \frac{1}{c_-} \E[(\Psi_\mu - \Psi_\nu) \ \L(\Psi_\mu - \Psi_\nu)].
\end{equation}
Since $\Psi_\mu = (\mu + \L)^{-1} f$, and using \eqref{e:spectr}, we can rewrite the expectation in the right-hand side above as
$$
\int \lambda \Ll( (\mu + \lambda)^{-1} - (\nu + \lambda)^{-1} \Rr)^2 \ \d e_f(\lambda) = \int \frac{\lambda (\nu - \mu)^2}{(\mu + \lambda)^2 (\nu + \lambda)^2} \ \d e_f(\lambda).
$$
Without loss of generality, we may assume that $\mu < \nu$. In this case, the integrand above is bounded by
$$
\frac{\lambda \nu^2}{\lambda^2 \nu^2} = \frac{1}{\lambda},
$$
which is integrable by \eqref{estimH-1}. Now, for every fixed $\lambda$, we have
$$
\frac{\lambda (\nu - \mu)^2}{(\mu + \lambda)^2 (\nu + \lambda)^2} \le \frac{\lambda \nu^2}{\lambda^4},
$$
which tends to $0$ as $\mu,\nu \to 0$. By the dominated convergence theorem, we thus obtain that the left-hand side of \eqref{e:stepcauchy} tends to $0$ as $\mu,\nu \to 0$, and thus that $(D\Psi_\mu)_{\mu > 0}$ is a Cauchy sequence in $L^2(\un{\O})$. Let us write $\chi$ for the limit. By definition, we have $\chi \in L^2_\na$. 

We now show that $\chi$ satisfies \eqref{e:station}. One can check that, as a consequence of the estimate \eqref{estimH-1} (and using \eqref{e:spectr}),
\begin{equation}
\label{estim2}
\sqrt{\mu} \Psi_\mu \xrightarrow[\mu \to 0]{L^2(\un{\O})} 0.
\end{equation}
For every square-integrable $g : \un{\O} \to \R$, we can write the weak formulation of \eqref{e:Psi}:
$$
\mu \E[g \ \Psi_\mu] + \E[D g \cdot A\z D\Psi_\mu] = \E[g \ D^* \cdot F].
$$
Using \eqref{estim2} and the Cauchy-Schwarz inequality, we get that $\mu \E[g \ \Psi_\mu]$ tends to $0$ as $\mu$ tends to $0$, and thus
\begin{equation}
\label{e:station-weak}
\E[D g \cdot A\z \chi] = \E[g \ D^* \cdot F].
\end{equation}
The left-hand side above is equal to $-\E[g \ D^* \cdot A\z \chi]$. Hence, the function $D^* \cdot A\z \chi + D^* \cdot F$ is orthogonal to every function in $L^2(\un{\O})$. It is thus equal to zero, and that is to say that $\chi$ satisfies \eqref{e:station}.

%Multiplying the above expression by $\Psi_\mu$ and taking expectations, we get 
%\begin{equation}
%\label{e:energy}	
%\mu \E[\Psi_\mu^2] + \E[D\Psi_\mu \cdot A\z D\Psi_\mu] = - \E[D \Psi_\mu \cdot F].
%\end{equation}
%As a consequence,
%$$
%\frac{1}{c_-} \|D\Psi_\mu\|_2^2 \le  \E[D\Psi_\mu \cdot A\z D\Psi_\mu] \le - \E[D \Psi_\mu \cdot F] \le \|D\Psi_\mu\|_2 \ \|F\|_2,
%$$
%and in particular, $\|D\Psi_\mu\|_2$ remains bounded uniformly over $\mu$. Hence, there exists a function $\chi \in L^2(\un{\O})$ such that $D\Psi_\mu$ converges to $\chi$ weakly in $L^2(\un{\O})$ along a subsequence (which we continue to write $\mu$ for simplicity). For every square-integrable $f : \un{\O} \to \R$, we can write the weak formulation of \eqref{e:Psi}:
%$$
%\mu \E[f \ \Psi_\mu] + \E[D f \cdot A\z D\Psi_\mu] = \E[f \ D^* \cdot F].
%$$
%By \eqref{e:energy} and the fact that $\|D\Psi_\mu\|_2$ remains bounded, we know that $\mu \E[\Psi_\mu^2]$ is bounded uniformly over $\mu$. As a consequence (and using the Cauchy-Schwarz inequality), we get that $\mu \E[f \ \Psi_\mu]$ tends to $0$ as $\mu$ tends to $0$, and thus
%\begin{equation}
%\label{e:station-weak}
%\E[D f \cdot A\z \chi] = \E[f \ D^* \cdot F].
%\end{equation}
%The left-hand side above is equal to $-\E[f \ D^* \cdot A\z \chi]$, and thus $\chi$ indeed satisfies \eqref{e:station}. In order to complete the proof of existence, there remains to see that $\chi \in L^2_\na$. In order to see this, we compute 
%\begin{equation}
%\label{e:stat1}
%\E[(D\Psi_\mu - \chi) \cdot A\z (D\Psi_\mu - \chi)] = \E[D\Psi_\mu \cdot A\z D\Psi_\mu] + \E[\chi \cdot A\z \chi] - 2 \E[D\Psi_\mu \cdot A\z \chi].
%\end{equation}
%Taking $f = \Psi_\mu$ in \eqref{e:station-weak} and passing to the limit, we get that 
%$$
%\lim_{\mu \to 0} \E[D\Psi_\mu \cdot A\z \chi] = \E[\chi \cdot A\z \chi] = -\E[\chi \cdot F].
%$$
%We have seen that $\mu \E[\Psi_\mu^2]$ tends to $0$ with $\mu$, so $\E[D\Psi_\mu \cdot A\z D\Psi_\mu]$ converges to $-\E[\chi \cdot F]$ by \eqref{e:energy},
%and we obtain that the left-hand side of \eqref{e:stat1} converges to $0$. This implies that $D\Psi_\mu$ converges to $\chi$ strongly in $L^2(\un{\O})$, and thus finishes the proof.

We now turn to uniqueness. By linearity, it suffices to show uniqueness for $F = 0$. Let $\chi \in L^2_\na$ satisfy 
\begin{equation}
\label{e:station0}
-D^* \cdot A\z \chi = 0.
\end{equation}
There exists a sequence of square-integrable $f_n : \un{\O} \to \R$ such that $D f_n$ converges to~$\chi$ in $L^2(\un{\O})$. By the weak formulation of \eqref{e:station0}, we have
$$
\E[Df_n \cdot A\z \chi] = 0.
$$
Passing to the limit, we get $\E[\chi \cdot A\z \chi] = 0$, which implies that $\chi = 0$, and thus finishes the proof.
\end{proof}

We now proceed to derive several convergence results that will be useful for our subsequent reasoning.

\begin{prop}[convergence of correctors]
\label{p:corrlim}
The function $\na \phi\z_\mu(0,\cdot)$ converges in $L^2(\un{\O})$, as $\mu$ tends to $0$, to the function $\na \phi\z(0,\cdot)$, while for every $e \in \B$, the function
$\na \phi^e_\mu(0,\cdot)$ converges in $L^2(\un{\O})$, as $\mu$ tends to $0$, to a function that we write $\na \phi^e(0,\cdot)$.
\end{prop}
\begin{proof}
The first part is classical (and is a minor adaptation of the proof of Theorem~\ref{t:station}, choosing $F = A\z \xi$). For the second part, in view of Lemma~\ref{l:one}, it suffices to study the convergence of $\na_2 G^\o_\mu(x,y)$, for any fixed $x, y \in \Z^d$ and $\omega \in \Omega$. We define
$$
\na_2 G^\o(x,y) = \int_0^{+\infty} \na_3 q^\o(t,x,y) \ \d t.
$$
Note that this definition makes sense even in dimension 2 by \eqref{parabpointgrad}, although $G^\o(x,y)$ itself does not. For the same reason, $\na_2 G^\o(x,y)$ is bounded uniformly over $\omega \in \Omega$. We get that
$$
\na_2 G^\o(x,y) - \na_2 G^\o_\mu(x,y) = \int_0^{+\infty} (1-e^{-\mu t}) \na_3 q^\o(t,x,y) \ \d t.
$$
This and \eqref{parabpointgrad} ensure that $\na_2 G^\o_\mu(x,y)$ converges to $\na_2 G^\o(x,y)$ as $\mu$ tends to $0$, uniformly over $\omega \in \Omega$. Using Lemma~\ref{l:one}, we thus obtain the convergence of $\na \phi^e_\mu$ in $L^2(\un{\O})$, as desired.
\end{proof}
\begin{rem}
\label{r:phie}
The proof shows that
\begin{equation}
\label{e:naphie}
\na \phi^e(0,\un{\o}) = \na \phi\z(0,\un{\o}) - \Ll( \na_1 \na_2 G^e(0,\un{e})\Rr)  C^e(\un{e}) (\xi + \na \phi\z(\un{e}) ).
\end{equation}
\end{rem}
\begin{prop}[convergence of full correction]
\label{p:defcoef}
Let $\psi_\mu : \un{\O} \to \R$ be defined by
$$
\psi_\mu = \sum_{e \in \B} (\phi^e_\mu - \phi\z_\mu)(0,\cdot).
$$
The gradient $D \psi_\mu$ converges in $L^2(\un{\O})$, as $\mu$ tends to $0$, to $\chi \in L^2_\na$ the unique solution of
\begin{equation}
\label{e:defcoef}
-D^* \cdot A\z \chi = D^* \cdot F,
\end{equation}
where $F \in L^2(\un{\O})$ is defined by
\begin{equation}
\label{defF}
F = \sum_{e \in \B} C^e (\xi + \na \phi^e)(0,\cdot).
\end{equation}
\end{prop}
\begin{rem}
Note that $\psi_\mu$ is well-defined, since $\td{\phi}^\B_\mu$ introduced in \eqref{deftdphi} is. Also, in the definition of $F$, note that the presence of the term $C^e$ ensures that every summand indexed by an edge $e$ such that $\un{e} \neq 0$ is actually equal to $0$. 
\end{rem}
\begin{proof}
Recall that, by definition, $\phi^e_\mu$ satisfies
$$
\mu \phi^e_\mu - \na^* \cdot A^e (\xi + \na \phi^e_\mu) = 0 \qquad (\text{in } \Z^d),
$$
and that $A^e = A\z+C^e$. As a consequence, we get
$$
\mu \phi^e_\mu - \na^* \cdot A\z (\xi + \na \phi^e_\mu) = \na^* \cdot C^e (\xi + \na \phi^e_\mu) \qquad (\text{in } \Z^d).
$$
Using also the definition of $\phi\z_\mu$, see \eqref{defphimu}, and recalling that we write $\ov{\phi}^e_\mu = \phi^e_\mu - \phi\z_\mu$, we obtain that
$$
\mu \ov{\phi}^e_\mu - \na^* \cdot A\z  \na \ov{\phi}^e_\mu = \na^* \cdot C^e (\xi + \na \phi^e_\mu) \qquad (\text{in } \Z^d),
$$
and thus
\begin{equation}
\label{e:defcoef1}
\mu \psi_\mu - \na^* \cdot A\z  \na \psi_\mu = \sum_{e \in \B} \na^* \cdot C^e (\xi + \na \phi^e_\mu) \qquad (\text{in } \Z^d).
\end{equation}
We let 
\begin{equation}
\label{defFmu}
F_\mu(x,\un{\o}) = \sum_{e \in \B} C^e (\xi + \na \phi^e_\mu)(x,\un{\o}).
\end{equation}
Note that $F_\mu$ is a stationary function, so we can identify it with the function
$$
\left\{
\begin{array}{lll}
\un{\O} & \to & \R^d \\
\un{\o} & \mapsto & F_\mu(0,\un{\o}).
\end{array}
\right.
$$
With this identification in mind, we can rewrite \eqref{e:defcoef1} as
\begin{equation}
\label{e:eqFmu}
\Ll( \mu - D^* \cdot A\z D \Rr) \psi_\mu = D^* \cdot F_\mu.
\end{equation}
By Proposition~\ref{p:corrlim}, the function $F_\mu$ converges in $L^2(\un{\O})$ to $F$ defined in \eqref{defF}.
We would like to argue as in the ``existence'' part of the proof of Theorem~\ref{t:station} to show that $D\psi_\mu$ converges to $\chi$ satisfying \eqref{e:defcoef}, but a difficulty arises since the right-hand side of \eqref{e:eqFmu} now depends on $\mu$. Let $\td{\psi}_\mu \in L^2(\un{\O})$ be the unique solution to
$$
\Ll( \mu - D^* \cdot A\z D \Rr) \td{\psi}_\mu = D^* \cdot F.
$$
By Theorem~\ref{t:station}, we know that $D\td{\psi}_\mu$ converges in $L^2(\un{\O})$ to $\chi \in L^2_\na$ the unique solution of \eqref{e:defcoef}. We now let $\ov{\psi}_\mu = \psi_\mu - \td{\psi}_\mu$, so that
$$
\Ll( \mu - D^* \cdot A\z D \Rr) \ov{\psi}_\mu = D^* \cdot (F_\mu - F).
$$
In order to conclude, we need to show that $D\ov{\psi}_\mu$ tends to $0$ in $L^2(\un{\O})$. Let us write $f_\mu = D^* \cdot (F_\mu - F)$. In view of \eqref{posit} and \eqref{e:spectr}, we have
$$
\|D\ov{\psi}_\mu\|_2^2 \le \frac{1}{c_-} \E[\ov{\psi}_\mu \ \L \ov{\psi}_\mu] = \frac{1}{c_-} \int\frac{\lambda}{(\mu + \lambda)^2} \ \d e_{f_\mu}(\lambda) \le \frac{1}{c_-} \int \frac{1}{\lambda} \ \d e_{f_\mu}(\lambda).
$$
By \eqref{estimH-1}, the last integral is bounded by
$$
\frac{\|F_\mu - F\|_2^2}{c_-},
$$
and since $F_\mu$ tends to $F$ in $L^2(\un{\O})$, this finishes the proof.
\end{proof}

%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{First-order expansion around $p = 0$}
\label{s:expansion0}
%\setcounter{equation}{0}

We are now ready to state and prove the first-order expansion around $p = 0$ of the homogenized matrix $\Ah\p$.

\begin{thm}[first-order expansion around $p=0$]
\label{t:main0}
There exists $a_1\z \in \R$ such that, as $p$ tends to $0$,
\begin{equation}
\label{e:main0}
\xi \cdot \Ah\p \xi = \xi \cdot \Ah\z \xi + p \ a_1\z + o(p).
\end{equation}
Moreover, the coefficient $a_1\z$ can be defined as the limit of $a_1\z(\mu)$ as $\mu$ tends to $0$, where $a_1\z(\mu)$ is given by
\begin{equation}
\label{defa1mu}
a_1\z(\mu) = \sum_{e \in \B} \Ll(\E[\xi \cdot A^e(\xi + \na {\phi}_\mu^e)] - \E[\xi \cdot A\z(\xi + \na {\phi}_\mu\z)] \Rr),
\end{equation}
and where we recall that $\phi_\mu^e$ satisfies \eqref{defphiE} with $E = \{e\}$, that $\phi_\mu\z$ satisfies \eqref{defphimu} with $p = 0$, and that the functions under the expectations in \eqref{defa1mu} are understood to be evaluated at $(0,\un{\o})$. Alternative characterizations of $a_1\z$ are given in Propositions~\ref{p:periodiz} and \ref{p:spatial-average} below.
\end{thm}
\begin{rem}
	The proof actually reveals that there exists an exponent $\ov{\eta} > 0$ depending on the ellipticity constants such that \eqref{e:main0} holds with $o(p)$ replaced by $o(p^{1+\ov{\eta}})$. We show in Theorem~\ref{t:error} that the error term is actually $o(p^{2-\eta})$ for every $\eta > 0$.
\end{rem}
\begin{proof}
For this proof, we fix
\begin{equation}
\label{fixmu}
\mu = p^{1+\eps},
\end{equation}
for some $\eps > 0$. Note that in order to avoid heavy notations, this dependence between $\mu$ and $p$ is kept implicit. For instance, and in view of Theorem~\ref{t:approx}, we may write without further comment that
\begin{equation}
\label{e:approx}
\Ll|\xi \cdot \Ah\p \xi - \E[\xi \cdot A\p(\xi + \na \phi_\mu\p)] \Rr|	= o\Ll(p\Rr) \qquad (p \to 0).
\end{equation}
For $\beta > 0$ given by Theorem~\ref{t:approxcorr}, we have
\begin{equation}
\label{e:approx2}
\Ll| \E[\xi \cdot A\p(\xi + \na \phi_\mu\p)]  - \E[\xi \cdot A\p(\xi + \na \td{\phi}_\mu\p)] \Rr| = O\Ll(p^{2-(1+\eps)(1-\beta)}\Rr)% \qquad (p \to 0)
.
\end{equation}
We fix the value of $\eps > 0$ in such a way that 
\begin{equation}
\label{e:fixeps}
2-(1+\eps)(1-\beta) > 1.
\end{equation}
Using the definition of $\td{\phi}_\mu$ introduced in \eqref{deftdphi} (with $E = E\p$), we can write
\begin{equation}
\label{e:1}
\E[\xi \cdot A\p(\xi + \na \td{\phi}_\mu\p)] = \E[\xi \cdot A\p(\xi + \na {\phi}_\mu\z)] + \E\Ll[ \sum_{e \in E\p} \xi \cdot A\p(\na \phi^e_\mu - \na \phi\z_\mu) \Rr].
\end{equation}
We analyse the two terms on the right-hand side separately, beginning with the first. Note that $A\p$ (evaluated at the origin) is equal to $A\z$ unless one of the edges attached to the origin belongs to $E\p$. Moreover, the event that two or more of these edges belong to $E\p$ has probability $O(p^2)$. As a consequence, and since $\E[|\na \phi_\mu\z|^2]$ is bounded uniformly over $\mu$, we have
\begin{multline}
\label{e:part1}
\E[\xi \cdot A\p(\xi + \na {\phi}_\mu\z)] \\
= \E[\xi \cdot A\z(\xi + \na {\phi}_\mu\z)] + p \sum_{e : \un{e} = 0} \Ll(\E[\xi \cdot A^e(\xi + \na {\phi}_\mu\z)] - \E[\xi \cdot A\z(\xi + \na {\phi}_\mu\z)] \Rr) + O(p^2).
\end{multline}
We now turn to the second term in the right-hand side of \eqref{e:1}, which we decompose into
\begin{equation}
\label{e:2}
\E\Ll[ \sum_{e \in E\p,  \un{e} = 0} \xi \cdot A\p(\na \phi^e_\mu - \na \phi\z_\mu) \Rr] + \E\Ll[ \sum_{e \in E\p,  \un{e} \neq 0} \xi \cdot A\p(\na \phi^e_\mu - \na \phi\z_\mu) \Rr].
\end{equation}
By Lemma~\ref{l:one} and \eqref{greenpoint2}, it is clear that $\E[|\na \phi^e_\mu|^2]$ is bounded uniformly over $\mu$. Hence, reasoning as above, we get
\begin{equation}
\label{e:3}
\E\Ll[ \sum_{e \in E\p,  \un{e} = 0} \xi \cdot A\p(\na \phi^e_\mu - \na \phi\z_\mu) \Rr] = p \sum_{e: \un{e} = 0} \E\Ll[ \xi \cdot A^e(\na \phi^e_\mu - \na \phi\z_\mu) \Rr] + O(p^2).
\end{equation}
We now consider the second term in \eqref{e:2}. For an edge $e$ such that $\un{e} \neq 0$, the event $e \in E\p$ is independent of the value of $A\p$ (at the origin). Hence, we can rewrite the second term in \eqref{e:2} as
$$
p \sum_{e : \un{e} \neq 0} \E[\xi \cdot A\p(\na \phi^e_\mu - \na \phi\z_\mu)],
$$
which itself can be rewritten as
$$
p \E[\xi \cdot A\p D\psi_\mu] - p \sum_{e : \un{e} = 0}\E[\xi \cdot A\p(\na \phi^e_\mu - \na \phi\z_\mu)],
$$
where $\psi_\mu$ was introduced in Proposition~\ref{p:defcoef}. Arguing as above, we see that, on the one hand,
$$
\sum_{e : \un{e} = 0}\E[\xi \cdot A\p(\na \phi^e_\mu - \na \phi\z_\mu)] = \sum_{e : \un{e} = 0}\E[\xi \cdot A\z(\na \phi^e_\mu - \na \phi\z_\mu)] + O(p),
$$
while on the other hand, since $D\psi_\mu$ remains bounded in $L^2(\un{\O})$ by Proposition~\ref{p:defcoef},
$$
\E[\xi \cdot A\p D\psi_\mu] = \E[\xi \cdot A\z D\psi_\mu] + O(p).
$$
We have thus shown that
\begin{multline*}
\E\Ll[ \sum_{e \in E\p,  \un{e} \neq 0} \xi \cdot A\p(\na \phi^e_\mu - \na \phi\z_\mu) \Rr] \\
= p\E[\xi \cdot A\z D\psi_\mu] - p\sum_{e : \un{e} = 0}\E[\xi \cdot A\z(\na \phi^e_\mu - \na \phi\z_\mu)] + O(p^2).
\end{multline*}
Combining this with \eqref{e:approx}, \eqref{e:approx2}, \eqref{e:1}, \eqref{e:part1}, \eqref{e:2} and \eqref{e:3}, we thus obtain
\begin{equation}
\label{e:presque}
%\E[\xi \cdot A\p(\xi + \na \td{\phi}_\mu\p)]
\xi \cdot \Ah\p \xi = \E[\xi \cdot A\z(\xi + \na {\phi}_\mu\z)] + p \ a_1\z(\mu) + o(p),
\end{equation}
where we introduced
\begin{multline}
\label{defa1zmu}
a_1\z(\mu)=  \sum_{e : \un{e} = 0} \Ll(\E[\xi \cdot A^e(\xi + \na {\phi}_\mu^e)] - \E[\xi \cdot A\z(\xi + \na {\phi}_\mu\z)] \Rr) + \\
\E[\xi \cdot A\z D\psi_\mu] - \sum_{e : \un{e} = 0}\E[\xi \cdot A\z(\na \phi^e_\mu - \na \phi\z_\mu)].
\end{multline}
Using the definition of $\psi_\mu$, one can observe that this definition coincides with that given in \eqref{defa1mu}. Using Theorem~\ref{t:approx} again, and recalling \eqref{fixmu}, we see that $\E[\xi \cdot A\z(\xi + \na {\phi}_\mu\z)] = \xi \cdot \Ah\z \xi + o(p)$. In order to conclude the proof, it thus suffices to show that $a_1\z(\mu)$ converges to a constant as $\mu$ tends to $0$. But this is a consequence of Propositions~\ref{p:corrlim} and \ref{p:defcoef}.
\end{proof}

%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Approximation by periodization}
\label{s:periodiz}
The aim of this section is to give alternative characterizations of $a_1\z$, based on computing correctors on periodizations of the medium. We recall that we write $B_n = \{-n, \ldots, n\}^d$, and we let
$\B_n = \{ e \in \B : \un{e} \in B_n\}$. 
We define the periodized environment $\on \in \un{\O}$ by letting, for every $e \in \B$,
$(\on)_e = \omega_{e+x}$, where $x$ is the unique element of $(2n+1)\Z^d$ such that $e+x \in \B_n$. By the Lax-Milgram lemma and the Poincaré inequality (or by basic linear-algebra considerations), for every $\un{\o} \in \un{\O}$, there exists a unique $B_n$-periodic function $\phi\z_n(\cdot, \on)$ with zero average over $B_n$ such that for every $x \in \Z^d$,
$$
-\na^* \cdot A\z (\xi + \na \phi\z_n)(x,\on) = 0.
$$
For the same reason, for every $e \in \B_n$, there exists a unique $B_n$-periodic function $\phi_n^e(\cdot,\on)$ with zero average over $B_n$ such that for every $x \in \Z^d$,
$$
-\na^* \cdot A^e (\xi + \na \phi^e_n)(x,\on) = 0.
$$
We write $\E_n[f]$ as a shorthand for $\E[f(\on)]$, whenever this is well-defined. As usual, if the function is of the form $f(x,\on)$, $x \in \Z^d$, we interpret $\E_n[f]$ to mean $\E[f(0,\on)]$.

%The purpose of this section is to prove the following result.
\begin{prop}[approximation of $a_1\z$ by periodization]
\label{p:periodiz}
The coefficient $a_1\z$ appearing in Theorem~\ref{t:main0} is the limit of $a_1\z(n)$ as $n$ tends to infinity, where
\begin{equation}
\label{defa1n}
a_1\z(n) = \sum_{e \in \B_n} \Ll( \E_n[\xi \cdot A^e (\xi + \na \phi^e_n)]  - \E_n[\xi \cdot A\z (\xi + \na \phi\z_n)] \Rr).
\end{equation}
\end{prop}
\begin{proof}
Defining
$$
\psi_n(\on) = \sum_{e \in \B_n} (\phi_n^e - \phi_n\z)(0,\on),
$$
we note that, in close resemblance with \eqref{defa1zmu},
\begin{multline}
\label{defa1zn}
a_1\z(n)=  \sum_{e : \un{e} = 0} \Ll(\E_n[\xi \cdot A^e(\xi + \na {\phi}_n^e)] - \E_n[\xi \cdot A\z(\xi + \na {\phi}_n\z)] \Rr) + \\
\E_n[\xi \cdot A\z D\psi_n] - \sum_{e : \un{e} = 0}\E_n[\xi \cdot A\z(\na \phi^e_n - \na \phi\z_n)].
\end{multline}
It thus suffices to prove the following three statements.
\begin{enumerate}
	\item The function $\un{\o} \mapsto \na \phi_n\z(0,\on)$ converges in $L^2(\un{\O})$ to $\na \phi\z(0,\cdot)$ as $n$ tends to infinity.
	\item The function $\un{\o} \mapsto \na \phi_n^e(0,\on)$ converges in $L^2(\un{\O})$ to $\na \phi^e(0,\cdot)$ as $n$ tends to infinity.
	\item The function $\un{\o} \mapsto D\psi_n(\on)$ converges in $L^2(\un{\O})$, as $n$ tends to infinity, to the function $\chi$ defined in Proposition~\ref{p:defcoef}.
\end{enumerate}
Part (1) is classical (see for instance \cite{capiof, boupia}), but it will be useful to recall a proof. We begin by observing that the function $\phi_n\z$ is stationary, in the sense that for every $\un{\o} \in \un{\O}$ and every $x \in \Z^d$, we have $\phi_n\z(x,\on) = \phi_n\z(0,\theta_x \ \on)$. As a consequence, we may identify $\phi_n\z$ with the function $\phi_n\z(0,\cdot)$. We then observe that
\begin{equation}
\label{period-energy}
\E_n\Ll[ D\phi_n\z \cdot A\z D\phi_n\z \Rr] = -\E_n\Ll[ D\phi_n\z \cdot A\z \xi \Rr],
\end{equation}
and thus, by the Cauchy-Schwarz inequality,
$$
c_- \E_n\Ll[|D\phi_n\z|^2\Rr] \le \E_n\Ll[ D\phi_n\z \cdot A\z D\phi_n\z \Rr] \le c_+ |\xi| \  \E_n\Ll[|D\phi_n\z|^2\Rr]^{1/2}.
$$
It follows from this inequality that $\E_n\Ll[|D\phi_n\z|^2\Rr]$ is bounded uniformly over $n$, and thus that $\un{\o} \mapsto D\phi_n\z(0,\on)$ converges weakly in $L^2(\un{\O})$ along a subsequence, to some $\td{\chi} \in L^2(\un{\O})$. For notational simplicity, we keep implicit the fact that we now consider $(D\phi_n\z)$ only along this subsequence. In order to prove part (1), it suffices to show that the weak convergence is actually strong convergence in $L^2(\un{\O})$, and that $\td{\chi} = \na \phi\z$. 

Note first that \eqref{period-energy} implies that (along the subsequence)
\begin{equation}
\label{csqenergy}
\lim_{n \to \infty} \E_n\Ll[ D\phi_n\z \cdot A\z D\phi_n\z \Rr] = -\E\Ll[\td{\chi} \cdot A\z \xi\Rr].
\end{equation}
Now, take any bounded function $f : \un{\O} \to \R$ that depends on the value of the environment only at a finite number of edges% (we may call this a \emph{local function})
. We have
$$
\E_n\Ll[Df \cdot A\z D\phi_n\z \Rr] = -\E_n\Ll[Df \cdot A\z \xi\Rr].
$$
Passing to the limit (along the subsequence), we get that
\begin{equation}
\label{weaktdchi}
\E\Ll[Df \cdot A\z \td{\chi} \Rr] = -\E\Ll[Df \cdot A\z \xi\Rr].
\end{equation}
The identity above can then be extended to arbitrary $f \in L^2(\un{\O})$. Replacing $f$ by $\phi_n\z$ in the above identity, and letting $n$ tend to infinity, we thus learn that
\begin{equation}
\label{weaktdchi2}
\E\Ll[\td{\chi} \cdot A\z \td{\chi} \Rr] = -\E\Ll[\td{\chi} \cdot A\z \xi\Rr].
\end{equation}
To see strong convegence of $D\phi_n$ to $\td{\chi}$ in $L^2(\un{\O})$, it now suffices to write 
\begin{multline*}
%\label{e:stat2}
\E_n\Ll[(D\phi_n - \td{\chi}) \cdot A\z (D\phi_n - \td{\chi}) \Rr] \\ = \E_n\Ll[D\phi_n \cdot A\z D\phi_n \Rr] + \E_n\Ll[\td{\chi} \cdot A\z \td{\chi} \Rr] - 2 \E_n\Ll[D\phi_n \cdot A\z \td{\chi} \Rr],
\end{multline*}
and observe that the right-hand side tends to zero by \eqref{csqenergy} and \eqref{weaktdchi2}. This ensures that $\td{\chi} \in L^2_\na$, and by \eqref{weaktdchi}, $\td{\chi}$ satisfies 
$$
-D^* \cdot A\z \td{\chi} = D^* \cdot A\z \xi.
$$
By Theorem~\ref{t:station}, there is a unique such $\td{\chi}$, and it is $\na \phi^\circ$ (see Proposition~\ref{p:corrlim}).

Part (3) can be obtained in a similar way, noting that the function $\psi_n$ is stationary, in the sense that $\psi_n(x,\on) = \psi_n(0, \theta_x \ \on)$.

For part (2), we cannot argue in the very same way, since $A^e$ is not stationary. We note instead that, letting $\ov{\phi}_n^e = \phi_n^e - \phi_n\z$, we have
$$
-\na^* \cdot A^e \na \ov{\phi}_n^e = \na^* \cdot C^e (\xi + \na \phi_n\z)
$$
at every point $(x,\on)$, with $x \in \Z^d$ and $\un{\o} \in \un{\O}$. We can then simply quote \cite[Lemma~A.2]{analeb2}, or argue as in part (1), with the difference that the arguments need to be carried out over the physical space $\Z^d$ instead of the space of environments. We get that $(x,\un{\o}) \mapsto \1_{B_n} \na \ov{\phi}_n^e(x,\on)$ converges in $L^2(\Z^d \times \un{\O})$ to $\na\ov{\phi}_\infty^e \in L^2(\Z^d\times \un{\O})$ as $n$ tends to infinity, where $\na \ov{\phi}_\infty^e$ satisfies
\begin{equation}
\label{e:ananth}	
-\na^* \cdot A^e \na \ov{\phi}_\infty^e = \na^* \cdot C^e (\xi + \na \phi\z) \quad  (\text{in } \Z^d \times \un{\O})
\end{equation}
(and the space $L^2(\Z^d \times \un{\O})$ is defined with respect to the measure obtained as the product of the counting measure over $\Z^d$ and $\P$). This identifies $\na \ov{\phi}_\infty^e$ as
$$
\na \ov{\phi}_\infty^e(x) = - \Ll(\na_1 \na_2 G^e(x,\un{e})\Rr)  C^e(\un{e}) (\xi + \na \phi\z(\un{e})).
$$
Comparing this with the formula for $\na \phi^e$ given in \eqref{e:naphie}, we see that we are done.
\end{proof}

The question arises as to whether the sum involving all $(\phi_n^e)_{e \in \B_n}$ in the formula defining $a\z_1(n)$ can be replaced by a spatial average involving only $(\phi_n^e)_{e :\un{e} = 0}$. This question is important since in practice, one would like to compute as few $\phi_n^e$'s as possible. We define
\begin{equation}
\label{defova}
\ov{a}_n(\on) = \sum_{x \in B_n} \sum_{e : \un{e} = 0} \Big( \xi \cdot A^{e}(\xi + \na \phi_n^e) -  \xi \cdot A\z(\xi + \na \phi_n\z) \Big)(x,\on).
\end{equation}
\begin{prop}[approximation of $a_1\z$ by periodization and spatial average]
\label{p:spatial-average}
The random variable $\un{\o} \mapsto \ov{a}_n(\on)$ converges in $L^1(\un{\O})$ to the random variable
$$
\sum_{e : \un{e} = 0} (\xi + \na \phi\z) \cdot C^e(\xi + \na \phi^e) \ (0,\cdot).
$$
Moreover, the coefficient $a_1\z$ given by Theorem~\ref{t:main0} satisfies
\begin{equation*}
%\label{e:spatial-average}
a_1\z = \sum_{e : \un{e} = 0} \E\Ll[(\xi + \na \phi\z) \cdot C^e(\xi + \na \phi^e) \Rr] = \sum_{e : \un{e} = 0} \lim_{n \to \infty} \E_n\Ll[ (\xi + \na \phi_n\z) \cdot C^e(\xi + \na \phi_n^e)  \Rr].
\end{equation*}
\end{prop}
\begin{proof}
Let $e$ be such that $\un{e} = 0$. %The computations below should be understood as calculated at $\on$, although we keep it implicit in the notation. 
By the definition of $\phi_n^e$ and $B_n$-periodicity,
$$
\sum_{x \in B_n} \xi \cdot A^e(\xi + \na \phi_n^e)(x,\on) = \sum_{x \in B_n} (\xi + \na \phi_n\z) \cdot A^e(\xi + \na \phi_n^e)(x,\on).
$$
We can decompose $A^e$ into $A\z + C^e$, and then observe that the definition of $\phi_n\z$ implies that
$$
\sum_{x \in B_n} (\xi + \na \phi_n\z) \cdot A\z(\xi + \na \phi_n^e)(x,\on) = \sum_{x \in B_n} (\xi + \na \phi_n\z) \cdot A\z \xi (x,\on).
$$
Since $A\z$ is a symmetric matrix, we have obtained that
\begin{equation}
\label{defovabis}
\ov{a}_n(\on) = \sum_{e : \un{e} = 0}(\xi + \na \phi_n\z) \cdot C^e(\xi + \na \phi_n^e)(0,\on).
\end{equation}
The convergence announced in the proposition then follows from the fact that $\un{\o} \mapsto \na \phi_n\z(0,\on)$ and $\un{\o} \mapsto \na \phi_n^e(0,\on)$ converge in $L^2(\un{\O})$ to $\na \phi\z(0,\cdot)$ and $\na \phi^e(0,\cdot)$ respectively. The second part of the proposition follows noting that $\E_n[\ov{a}_n] = a_1\z(n)$, see \eqref{defa1n}.
\end{proof}


%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{First-order expansion around every point}
\label{s:expansion}
%\setcounter{equation}{0}

The aim of this section is to generalize Theorem~\ref{t:main0} by giving a first-order expansion of $\Ah\ppp$ as $p$ tends to $0$, for every $\ov{p} \in [0,1]$. 

In order to state the result, it is convenient to introduce some new notation. For every $\ov{p} \in [0,1]$ and every $e \in \B$, we let $A\ppep = A^{E\pp \cup \{e\}}$ (recall that $A^E$ was defined in \eqref{defAE}), $\phi_\mu\ppep = \phi_\mu^{E\pp \cup \{e\}}$ (defined in \eqref{defphiE}), and so on. Similarly, we define $A\ppem = A^{E\pp \setminus \{e\}}$,  $\phi_\mu\ppem = \phi_\mu^{E\pp \setminus \{e\}}$, and so on.








\begin{thm}[first-order expansion around every point]
\label{t:main}
Let $\ov{p} \in [0,1]$. There exists $a_1\pp \in \R$ such that, as $p$ tends to $0$,
\begin{equation}
\label{e:main}
\xi \cdot \Ah\ppp \xi = \xi \cdot \Ah\pp \xi + p \ a_1\pp + o(p).
\end{equation}
Moreover, the coefficient $a_1\pp$ can be defined as the limit of $a_1\pp(\mu)$ as $\mu$ tends to $0$, where $a_1\pp(\mu)$ is given by
\begin{equation}
\label{defa1mupp}
a_1\pp(\mu) = \sum_{e \in \B} \Ll(\E[\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppep)] - \E[\xi \cdot A\ppem(\xi + \na {\phi}_\mu\ppem)] \Rr).
\end{equation}
Alternative characterizations of $a_1\pp$ are given in Remark~\ref{r:1} below.
\end{thm}


One may attempt to prove this result by defining a coupling of the Bernoulli random variables such that if $p_1 \le p_2$, then $E^{(p_1)} \subset E^{(p_2)}$, and then try to adapt the proof of Theorem~\ref{t:main0}. This approach leads to a serious difficulty when trying to estimate the left-hand side of \eqref{e:thestep}. Indeed, this estimate crucially relies on the fact that the non-perturbed environment is independent of the events $e \in E_\mu\p$. A naive adaptation of this proof to cover the expansion of $\Ah\ppp$ as $p$ tends to $0$ would thus require that the environment $\o\pp$ be independent of the events $e \in E_\mu\ppp \setminus E_\mu\pp$, and of course, this independence does not hold. 

This problem can be overcome with a control of higher moments of the gradients of the Green functions than that given by Proposition~\ref{p:greenaverage}. Such results have been obtained very recently in \cite{marott}. We will describe and use these results in the next section, but for the problem at hand, a more direct proof can be obtained. 


\begin{proof}[Proof of Theorem~\ref{t:main}]
It comes out of the above discussion that the decision as to which edges are to be perturbed should be taken independently of the environment. We thus introduce new independent Bernoulli random variables $(\td{b}_e\p)_{e \in \B}$ of parameter $p \in [0,1]$, independent of everything else, defined on the same probability space (possibly enlarging it)%, and we now understand that $\un{\o} = (\o_e,{\o}^{(1)}_e,(b_e\p,\td{b}_e\p)_{p \in [0,1]})_{e \in \B}$
. For $\ov{p} < 1$ and $p \in [0,1-\ov{p}]$, we further define
$$
\td{p} = \frac{p}{1-\ov{p}} \qquad \text{and} \qquad \td{\o}_e\ppvp = (1- \td{b}_e\tp) \ \omega\pp_e + \td{b}_e\tp \ \omega^{(1)}.
$$
Recalling that we write $\nu\p$ for the law of $\o_e\p$, we see that the law of $\td{\o}_e\ppvp$ is 
\begin{multline*}
(1-\td{p}) \nu\pp + \td{p} \nu^{(1)} = (1-\td{p})\Ll[ (1-\ov{p}) \nu^{(0)} + \ov{p} \nu^1 \Rr] + \td{p} \nu^{(1)} \\ = (1-\ov{p} - p) \nu^{(0)} + (\ov{p} + p) \nu^{(1)} = \nu\ppp.
\end{multline*}
In other words, the law of $\td{\o}_e\ppvp$ is that of $\o_e\ppp$, and as a consequence, the law of $\td{\o}\ppvp := (\td{\o}_e\ppvp)_{e \in \B}$ is that of $\o\ppp$.

We can apply Theorem~\ref{t:main0} for the environment $\td{\o}\ppvp$, seen as a Bernoulli perturbation of $\o\pp$, with the perturbed environment taken from $\o^{(1)}$ and the perturbation parameter being $\td{p}$. Since the homogenized matrix of $\td{\o}\ppvp$ is that of $\o\ppp$, we get that as $\td{p} \ge 0$ tends to $0$,
$$
\xi \cdot \Ah\ppp \xi = \xi \cdot \Ah\pp \xi + \td{p} \ \td{a}_1\pp + o(\td{p}),
$$
where $\td{a}_1\pp$ is the limit as $\mu$ tends to $0$ of $\td{a}_1\pp(\mu)$ given by
$$
\td{a}_1\pp(\mu) = \sum_{e \in \B} \Ll(\E[\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppep)] - \E[\xi \cdot A\pp(\xi + \na {\phi}_\mu\pp)] \Rr).
$$
We now see that the quantity
$$
\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppep) - \xi \cdot A\pp(\xi + \na {\phi}_\mu\pp)
$$
is equal to $0$ if $e \in E\pp$, and otherwise is equal to 
$$
\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppep) - \xi \cdot A\ppem(\xi + \na {\phi}_\mu\ppem).
$$
Since this last quantity is independent of the event $e \in E\pp$, we obtain that
\begin{multline}
\E\Ll[\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppep) - \xi \cdot A\pp(\xi + \na {\phi}_\mu\pp)\Rr] \\
= \P\Ll[e \notin E\pp\Rr] \ \E\Ll[ \xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppep) - \xi \cdot A\ppem(\xi + \na {\phi}_\mu\ppem) \Rr].
\end{multline}
Since $\P\Ll[e \notin E\pp\Rr] = 1-\ov{p}$, we obtain the desired result for $p \ge 0$ (i.e. for the right derivative of $\xi \cdot \Ah^{(\cdot)} \xi$). 

A similar reasoning can be followed if $\ov{p} \in (0,1]$ and $p \in [-\ov{p},0]$, by letting
$$
\td{p} = \frac{-p}{\ov{p}} \qquad \text{and} \qquad \td{\o}_e\ppvp = (1- \td{b}_e\tp) \ \omega\pp_e + \td{b}_e\tp \ \omega^{(0)}.
$$
The reasoning leads to the same formula for the left derivative, so the proof is complete.
\end{proof}

\begin{rem}
\label{r:1}
From the construction in the proof of Theorem~\ref{t:main} and Proposition~\ref{p:periodiz}, one can give an alternative characterization of the coefficient $a_1\pp$ appearing in Theorem~\ref{t:main}. Indeed, it is the limit of $a_1(n)$ as $n$ tends to infinity, for
$$
a_1(n) = \sum_{e \in \B_n} \Ll( \E_n\Ll[\xi \cdot A\ppep(\xi + \na \phi_n\ppep)\Rr] - \E_n\Ll[\xi \cdot A\ppem(\xi + \na \phi_n\ppem)\Rr]\Rr),
$$
where $\phi_n\ppep$ and $\phi_n\ppem$ are defined as $\phi_n\z$, but with respect to $A\ppep$ and $A\ppem$ respectively instead of $A\z$. 

Similarly, using Proposition~\ref{p:spatial-average}, we can obtain yet another characterization of $a_1\pp$, namely
\begin{equation}
\label{e:a1ppe}
a_1\pp =\sum_{e : \un{e} = 0} \E\Ll[ (\xi + \na \phi\ppem) \cdot C^e (\xi + \na \phi\ppep)\Rr],
\end{equation}
where $\na \phi\ppem(0,\cdot)$ and $\na \phi\ppep(0,\cdot)$ are the $L^2(\un{\O})$-limits of $\na \phi_n\ppem(0,\cdot)$ and $\na \phi_n\ppep(0,\cdot)$ respectively, as $n$ tends to infinity. Moreover,
\begin{equation}
\label{e:a1ppbis}
a_1\pp = \sum_{e : \un{e} = 0} \lim_{n \to \infty} \E_n\Ll[ (\xi + \na \phi_n\ppem) \cdot C^e (\xi + \na \phi_n\ppep) \Rr].
% = \lim_{n \to \infty} \sum_{x \in B_n} \sum_{e : \un{e} = 0} \E \Ll[ \Big(\xi \cdot A\ppep(\xi + \na \phi_n\ppep) -  \xi \cdot A\ppem(\xi + \na \phi_n\ppem)\Big)(x,\on) \Rr]
\end{equation}
\end{rem}

%
%
%
%
%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
%
%
%
\section{Refined control of the error term}
\label{s:error}
%\setcounter{equation}{0}

The aim of this section is to obtain a sharper control of the error term $o(p)$ appearing in \eqref{e:main} of Theorem~\ref{t:main}. As is clear from the proof of Theorem~\ref{t:main}, in order to achieve this, it is in fact sufficient to control the error term in Theorem~\ref{t:main0}. The fact that Theorem~\ref{t:main0} gives only a weak control of the error term lies in Theorem~\ref{t:approxcorr}. The weak point in the argument leading to Theorem~\ref{t:approxcorr} is in the estimation of the left-hand side of \eqref{e:thestep}, where we crudely estimated the second mixed derivatives of $G_\mu\p$ using the pointwise inequality \eqref{greenpoint2} on the gradient of $G_\mu\p$. Although Proposition~\ref{p:greenaverage} shows that the $L^1$ norm of the second mixed derivative of $G_\mu\p$ has the same spatial decay as the homogeneous case, this is of no use when estimating the left-hand side of \eqref{e:thestep} because we have no control on the dependence between the second derivatives of $G_\mu\p$ and the events $e \in E_\mu\p$. 

This problem can however be overcome with a control of higher moments of the second derivatives of the Green function. Such results have been obtained very recently in the remarkable work \cite{marott}.
\begin{thm}[\cite{marott}]
\label{t:marott}
For every $q \ge 1$, there exists a finite $\C_q$ depending only on $q$, $d$ and the ellipticity constants $c_-, c_+$ (but otherwise not on the law of $\omega$, as long as it is a product measure as assumed throughout) such that for every $x \in \Z^d$ and $\mu \ge 0$,
\begin{equation}
\label{e:marott}
\E\Ll[|\na_1 \na_2 G_\mu(0,x)|^{q}\Rr]^{1/q} \le \frac{\C_q}{(1 \vee |x|)^d}.
\end{equation}
\end{thm}
\begin{rem}
Striclty speaking, \cite[Theorem~1]{marott} corresponds to this result for $\mu = 0$. The proof can however be adapted with minor modifications to yield Theorem~\ref{t:marott}. Let us describe briefly how. \cite[Lemma~4]{marott} needs no change. \cite[Lemma~6]{marott} consists of two parts. The proof of the first part can be easily adapted to yield the result with $G$ replaced by $G_\mu$ (actually, it provides an alternative route to the periodization argument used there), while the second part was proved for $G_\mu$ in the first place (that is, in \cite{glotto}). The first part of \cite[Lemma~5]{marott} is all what is needed for our purpose, and it remains true with $G$ replaced by $G_\mu$ (uniformly over $\mu$). Indeed, the formulas for the derivatives of the Green function with respect to $\omega_e$ appearing in steps 1 and 2 of the proof, as e.g. \cite[(48)-(50)]{marott}, remain valid provided $G$ is replaced by $G_\mu$ everywhere. Step 3 is a consequence of \cite[Lemma~6]{marott}, while step 4 follows from previous steps, and concludes the proof of the part of the lemma we are interested in. Finally, the proof of \cite[Theorem~1]{marott} follows from the results we just reviewed (again providing an alternative route to the periodization argument proposed there). The fact that the constant $\C_q$ in \eqref{e:marott} can be chosen as a function of $q$, $d$ and the ellipticity constants only boils down to the fact that the constant $\rho$ in the weak logarithmic Sobolev inequality defined in \cite[(2)]{marott} can be chosen as a function of $c_-$ and $c_+$ only.
\end{rem}


\begin{thm}[sharp control of the error]
\label{t:error}
Let $\ov{p} \in [0,1]$ and $a_1\pp \in \R$ be given by Theorem~\ref{t:main}. For every $\eta >0$ and as $p$ tends to $0$, 
\begin{equation}
\label{e:error}
\xi \cdot \Ah\ppp \xi = \xi \cdot \Ah\pp \xi + p \ a_1\pp + o(p^{2-\eta}).
\end{equation}
%Moreover, the coefficient $a_1\pp$ can be defined as the limit of $a_1\pp(\mu)$ as $\mu$ tends to $0$, where $a_1\pp(\mu)$ is given by
%\begin{equation}
%\label{defa1mupp}
%a_1\pp(\mu) = \sum_{e \in \B} \Ll(\E[\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppep)] - \E[\xi \cdot A\ppem(\xi + \na {\phi}_\mu\ppem)] \Rr).
%\end{equation}
%Alternative characterizations of $a_1\pp$ are given in Remarks~\ref{r:1} and \ref{r:2} below.
\end{thm}

As was said before, it suffices to prove Theorem~\ref{t:error} for $\ov{p} = 0$, and the key step for doing so is to prove the following result.

\begin{thm}[sharp linear approximation of the corrector] 
\label{t:sharp}
For every $\eta > 0$, there exist constants $c$, $\gamma > 0$ such that for every $\mu \in (0,1/2]$ and $p \in [0,1]$,
$$
\E\Ll[\Ll|\na \phi\p_\mu - \na \td{\phi}\p_\mu \Rr|\Rr] \le c p^{2-\eta} \mu^{-\eta} + e^{-\mu^{-\gamma}}.
$$
\end{thm}
We start with a result whose purpose is to replace Lemma~\ref{l:greenperturbed}.
\begin{lem}[sharp averaged control of the perturbed Green function]
There exists a finite constant $c$ (depending only on the ellipticity constants and on the dimension) such that for every $q \ge 1$, $\mu \in (0,1/2]$ and $e,e' \in \B$,
$$
\E\Ll[ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})|^{q}  \Rr]^{1/q} \le c \   \frac{\C_q}{(1\vee |\un{e}-\un{e}'|)^d},
$$
where $\C_q$ is the constant of Theorem~\ref{t:marott}.
\label{l:sharpgreenperturbed}
\end{lem}
\begin{proof}
Recall from the proof of Lemma~\ref{l:greenperturbed} that  $\ov{G}^e_\mu = G^e_\mu - G\z_\mu$. Since by Theorem~\ref{t:marott},
\begin{equation}
\label{e:sharpgreen}
\E\Ll[ | \na_1 \na_2 G\z_\mu(\un{e}',\un{e})|^{q}  \Rr]^{1/q} \le \frac{\C_q}{(1\vee |\un{e}-\un{e}'|)^d},
\end{equation}
it suffices to prove that
$$
\E\Ll[ | \na_1 \na_2 \ov{G}^e_\mu(\un{e}',\un{e})|^{q}  \Rr]^{1/q} \le c \ \frac{\C_q}{(1\vee |\un{e}-\un{e}'|)^d}.
$$
Recall from \eqref{relovG} that
$$
\na_1 \na_2\ov{G}^e_\mu(\un{e}',\un{e}) = C^e(\un{e}) \Ll(\na_1 \na_2 G^e_\mu(\un{e},\un{e})\Rr) \ \Ll(\na_1 \na_2G\z_\mu(\un{e}',\un{e})\Rr).
$$
Since, by \eqref{greenpoint2},
$$
\Ll|  C^e(\un{e}) \na_1 \na_2 G^e_\mu(\un{e},\un{e}) \Rr|
$$
is bounded uniformly over $\mu$ and $\un{\o}$, the result follows from \eqref{e:sharpgreen}.
\end{proof}
\begin{proof}[Proof of Theorem~\ref{t:sharp}]
For the beginning of the proof, we can follow the same reasoning as in the proof of Theorem~\ref{t:approxcorr}, up to the point when we arrive at the estimation of the left-hand side of \eqref{e:thestep}, that is,
$$
\E\Ll[ \sum_{e\neq e' \in  E\p_\mu}  | \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})| \Rr],
$$
where we recall that $E_\mu\p = E\p \cap B_{\mu^{-\gamma}}$, and $\gamma > 1/2$. We rewrite slightly this expectation as
\begin{equation}
\label{err1}
\sum_{e \neq e' \in B_{\mu^{-\gamma}}} \E\Ll[  \1_{e,e' \in E\p} | \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})| \Rr].
\end{equation}
Let $q > 3$ and $\zeta=1-3/q$. We apply Hölder's inequality with exponents $(\zeta^{-1},q,q,q)$ to bound each summand in \eqref{err1} by
\begin{multline}
\label{err2}
\P[e,e' \in E\p]^{\zeta} \ \E\Ll[ | \na_1 \na_2 G\p_\mu(0,\un{e}')|^{q}\Rr]^{1/q} \\
\E\Ll[ \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})|^{q}\Rr]^{1/q} \ \E\Ll[ |\xi + \na \phi\z_\mu(\un{e})|^{q}\Rr]^{1/q}.
\end{multline}
Since we only consider summands for which $e \neq e'$, we have
$$
\P[e,e' \in E\p]^{\zeta} = p^{2 \zeta}.
$$
By Theorem~\ref{t:marott} and Lemma~\ref{l:sharpgreenperturbed}, the product of the second and third terms in \eqref{err2} is bounded by
$$
c \ \frac{(\C_q)^2}{(1 \vee |\un{e}'|)^d \ (1 \vee |\un{e} - \un{e}'|)^d},
$$
while, by Theorem~\ref{t:approx}, the last term in \eqref{err2} is bounded by some power of $\log(\mu^{-1})$, say $\log^r(\mu^{-1})$. We thus obtain that the sum in \eqref{err1} is bounded by
$$
c \ p^{2\zeta} \log^r(\mu^{-1})  \sum_{e \neq e' \in B_{\mu^{-\gamma}}} \frac{(\C_q)^2}{(1 \vee |\un{e}'|)^d \ (1 \vee |\un{e} - \un{e}'|)^d}.
$$
By comparing this sum to an integral, we get that the whole expression above is bounded, up to a constant, by 
$p^{2\zeta} \mu^{-\gamma \eta}$,
for any $\eta > 0$. Recalling that $\zeta = 1-3/q$ and that $q$ can be chosen arbitrarily large, this concludes the proof.
\end{proof}
%
%
%
\begin{proof}[Proof of Theorem~\ref{t:error}] As noted before, it suffices to prove the theorem for $\ov{p} = 0$. The proof follows closely that of Theorem~\ref{t:main0}. Let $\eta > 0$. Instead of \eqref{fixmu}, we fix $\mu = p^2$. With this choice, Theorem~\ref{t:approx} implies that the left-hand side of \eqref{e:approx} is $O(p^{2-\eta})$, while by Theorem~\ref{t:sharp}, the left-hand side of \eqref{e:approx2} is $O(p^{2-3\eta})$. We can then follow the proof of Theorem~\ref{t:main0} without any change until \eqref{e:presque}, whose error term is now $O(p^{2-3\eta})$ instead of $o(p)$. The proof is then concluded in the same way, noting that Theorem~\ref{t:approx} ensures that $\E[\xi \cdot A\z(\xi + \na {\phi}_\mu\z)] = \xi \cdot \Ah\z \xi + O(p^{2-\eta})$.
\end{proof}



%For every bounded measurable function $f : [c_-,c_+] \to \R_+$, we define the \emph{entropy} of $f$ with respect to the law of $\omega_e$ as
%$$
%\ent(f) = \E[f(\o_e) \log f(\o_e) ] - \E[f(\o_e)] \log \E[f(\o_e)].
%$$
%%Recall that under our present assumptions, the random variables $(\o_e)_{e \in \B}$ all have the same law, so this definition does not depend on the edge $e$. 

%We say that the law of $\o_e$ satisfies a \emph{continuous log-Sobolev inequality with constant} $\rho > 0$ if for every continuously differentiable function $f : [c_-, c_+] \to \R$, we have
%$$
%\ent(f^2) \le \frac{1}{\rho} \E[(f'(\omega_e))^2].
%$$
%We say that the law of $\o_e$ satisfies a \emph{discrete log-Sobolev inequality with constant}~$\rho > 0$ if for every bounded measurable function $f : [c_-, c_+] \to \R$, we have
%$$
%\ent(f^2) \le \frac{1}{\rho} \E[(f(\omega_e) - \E[f(\omega_e)])^2].
%$$
%As is well-known, a very convenient feature of the log-Sobolev inequalities is that they can be tensorized to yield inequalities for functions defined on the product space $\Omega$ (see \cite[Theorem~4.4]{guizeg}). Such inequalities form one of the building blocks of the approach of \cite{marott}. We only quote the result of \cite{marott} that concerns the second mixed derivative of the Green function, but similar results were obtained for the first derivative as well.
%\begin{thm}[\cite{marott}]
%\label{t:marott}
%Assume that the law of $\o_e$ satisfies either the continuous or the discrete log-Sobolev inequality with constant $\rho > 0$. For every $q \ge 1$, there exists a finite $C = C(d,c_-,c_+,\rho,q)$ such that for every $x \in \Z^d$ and $\mu \ge 0$,
%$$
%\E\Ll[|\na_1 \na_2 G_\mu(0,x)|^{q}\Rr]^{1/q} \le \frac{C}{(1 \vee |x|)^d}.
%$$
%\end{thm}
%\begin{rem}
%Striclty speaking, \cite[Theorem~1]{marott} corresponds to this result for $\mu = 0$. The proof can however be adapted with minor modifications to yield Theorem~\ref{t:marott}. Let us describe briefly how. \cite[Lemma~1]{marott} needs no change. \cite[Lemma~3]{marott} consists of two parts. The proof of the first part can be easily adapted to yield the result with $G$ replaced by $G_\mu$ (actually, it provides an alternative route to the periodization argument used there), while the second part was proved for $G_\mu$ in the first place (that is, in \cite{glotto}). The first part of \cite[Lemma~2]{marott} is all what is needed for our purpose, and it remains true with $G$ replaced by $G_\mu$ (uniformly over $\mu$). Indeed, the formulas for the derivatives of the Green function with respect to $\omega_e$ appearing in steps 1 to 4 of the proof, as e.g. \cite[(39)-(41)]{marott}, remain valid provided $G$ is replaced by $G_\mu$ everywhere. Step 5 is a consequence of \cite[Lemma~3]{marott}, while step 6 follows from previous steps, and concludes the proof of the part of the lemma we are interested in. Finally, the proof of \cite[Theorem~1]{marott} follows from the results we just reviewed (again providing an alternative route to the periodization argument proposed there).
%\end{rem}

%%Theorem~\ref{t:marott} is particularly interesting since, for instance, every law whose support is a finite set satisfies a discrete log-Sobolev inequality. 

%%\medskip

%The results we will prove in this section are conditional on an assumption that we now introduce, which is related to the conclusion of Theorem~\ref{t:marott}. For every $p \in [0,1]$ and every $q \ge 1$, let $\C_q\p \in (0,+\infty]$ be defined by
%\begin{equation}
%\label{defCqp}
%\C_q\p = \sup_{x \in \Z^d, \mu \ge 0} (1 \vee |x|)^d \ \E\Ll[|\na_1 \na_2 G\p_\mu(0,x)|^{q}\Rr]^{1/q}.
%\end{equation}
%Let $\ov{p} \in [0,1]$. We say that \emph{assumption $(\H\pp)$ holds} if for every $q \ge 1$, there exists a neighbourhood of $\ov{p}$ in $[0,1]$ such that $\C_q\p$ is bounded uniformly over $p$ in this neighbourhood.

%%In order to state the main result of this section, it is convenient to introduce some new notation. For every $\ov{p} \in [0,1]$ and every $e \in \B$, we let $A\ppep = A^{E\pp \cup \{e\}}$ (recall that $A^E$ was defined in \eqref{defAE}), $\phi_\mu\ppep = \phi_\mu^{E\pp \cup \{e\}}$ (defined in \eqref{defphiE}), and so on. Similarly, we define $A\ppem = A^{E\pp \setminus \{e\}}$,  $\phi_\mu\ppem = \phi_\mu^{E\pp \setminus \{e\}}$, and so on.

%\begin{thm}[sharp control of the error]
%\label{t:error}
%Let $\ov{p} \in [0,1]$ and $a_1\pp \in \R$ be given by Theorem~\ref{t:main}. If assumption $(\H\pp)$ holds, then for every $\eta >0$, 
%\begin{equation}
%\label{e:error}
%\xi \cdot \Ah\ppp \xi = \xi \cdot \Ah\pp \xi + p \ a_1\pp + o(p^{2-\eta}).
%\end{equation}
%%Moreover, the coefficient $a_1\pp$ can be defined as the limit of $a_1\pp(\mu)$ as $\mu$ tends to $0$, where $a_1\pp(\mu)$ is given by
%%\begin{equation}
%%\label{defa1mupp}
%%a_1\pp(\mu) = \sum_{e \in \B} \Ll(\E[\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppep)] - \E[\xi \cdot A\ppem(\xi + \na {\phi}_\mu\ppem)] \Rr).
%%\end{equation}
%%Alternative characterizations of $a_1\pp$ are given in Remarks~\ref{r:1} and \ref{r:2} below.
%\end{thm}

%Before we proceed to the proof of Theorem~\ref{t:error}, we comment on the relations between assumption $(\H\pp)$ and Theorem~\ref{t:marott}. We see from Theorem~\ref{t:marott} that in order to ensure that $(\H\pp)$ holds, it suffices to show that the law of $\o_e\p$ satisfies a continuous or discrete log-Sobolev inequality with constant $\rho\p$, and that $\rho\p$ is uniformly bounded away from $0$ in a neighbourhood of $\ov{p}$.

%For many distributions of interest, as for instance all those with a finite support, the continuous log-Sobolev inequality fails to hold. We thus turn our attention to the discrete version of the inequality. The set of distributions satisfying a discrete log-Sobolev inequality is well-understood. Indeed, the law of $\o_e$ satisfies a discrete log-Sobolev inequality if and only if it has finite support. In this case, the optimal constant in the log-Sobolev inequality is explicit, and given by
%\begin{equation}
%\label{logsobconstant}
%\frac{1-2\pi_*}{\log[(1-\pi_*)/\pi_*]},
%\end{equation}
%where $\pi_* = \min \P[\o_e = x]$, and the minimum is taken over all $x$ ranging in the (finite) support of the law of $\o_e$ \cite[Theorem~A.1]{diasal} (here, we assumed that the law of $\o_e$ is not concentrated on a single point, in which case $\pi_* \le 1/2$ ; the quantity in \eqref{logsobconstant} can be extended by continuity to $\pi_* = 1/2$ and takes the value $1/2$ there).

%The constant in \eqref{logsobconstant} is bounded away from $0$ as soon as $\pi_*$ itself remains bounded away from $0$. By Theorem~\ref{t:marott}, this shows that assumption $(\H\pp)$ holds as soon as the laws of $\o_e$ and ${\o}^{(1)}_e$ both have finite support and $\ov{p} \notin \{0,1\}$. If the laws of $\o_e$ and ${\o}^{(1)}_e$ both have finite support, then assumption $(\H\z)$ holds as soon as the support of $\o^{(1)}_e$ is included in that of $\o_e$ (we write $(H\z)$ for $(H^{(0)})$). The same statement with $\o_e$ and $\o^{(1)}_e$ interchanged holds concerning assumption $(\H^{(1)})$. 

%Note for instance that the argument above cannot show that assumption $(\H\z)$ holds in the case where the environments are (distinct) constants! Yet, the idea that assumption $(\H\z)$ really breaks down in this case is highly doubtful. It is thus very likely that the problem we are encountering here is an artefact of the proof techniques. In fact, I believe that assumption $(H\pp)$ holds for every $\ov{p}$ without any further assumptions than those introduced in section~\ref{s:defs}.




%\medskip

%We now turn to the proof of Theorem~\ref{t:error}. 
%%We begin by introducing a coupling of the Bernoulli random variables $(b\p_e)_{e \in \B}$. We let $(U_e)_{e \in \B}$ be independent random variables, whose common distribution is uniform over $[0,1]$. We now assume that the Bernoulli random variables are defined by
%%$$
%%b\p_e = \1_{\{U_e \le p\}}.
%%$$
%%In this way, for every $p$, the random variables $(b\p_e)_{e \in \B}$ are independent and following a Bernoulli distribution with parameter $p$, as was assumed throughout. Moreover, and this is the point of the coupling, if $p_1 \le p_2$, then $b^{(p_1)}_e \le b^{(p_2)}_e$ for every $e \in \B$, or in other words, we have $E^{(p_1)} \subset E^{(p_2)}$.
%%
%As was said before, it suffices to prove it for $\ov{p} = 0$, and the key step for doing so is to prove the following result.

%\begin{thm}[sharp linear approximation of the corrector] 
%\label{t:sharp}
%If assumption $(\H\z)$ holds, then for every $\eta > 0$, there exist constants $c$, $\gamma > 0$ such that for every $\mu \in (0,1/2]$ and $p \in [0,1]$,
%$$
%\E\Ll[\Ll|\na \phi\p_\mu - \na \td{\phi}\p_\mu \Rr|\Rr] \le c p^{2-\eta} \mu^{-\eta} + e^{-\mu^{-\gamma}}.
%$$
%\end{thm}
%We begin the proof with a result whose purpose is to replace Lemma~\ref{l:greenperturbed}.
%\begin{lem}[sharp averaged control of the perturbed Green function]
%There exists a constant $c < +\infty$ (depending only on the ellipticity constants and on the dimension) such that for every $q \ge 1$, $\mu \in (0,1/2]$ and $e,e' \in \B$,
%$$
%\E\Ll[ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})|^{q}  \Rr]^{1/q} \le c \ \frac{\C_q\z}{(1\vee |\un{e}-\un{e}'|)^d}.
%$$
%\label{l:sharpgreenperturbed}
%\end{lem}
%\begin{proof}
%Recall from the proof of Lemma~\ref{l:greenperturbed} that  $\ov{G}^e_\mu = G^e_\mu - G\z_\mu$. Since by definition of $\C_q\z$, 
%\begin{equation}
%\label{e:sharpgreen}
%\E\Ll[ | \na_1 \na_2 G\z_\mu(\un{e}',\un{e})|^{q}  \Rr]^{1/q} \le \frac{\C_q\z}{(1\vee |\un{e}-\un{e}'|)^d},
%\end{equation}
%it suffices to prove that
%$$
%\E\Ll[ | \na_1 \na_2 \ov{G}^e_\mu(\un{e}',\un{e})|^{q}  \Rr]^{1/q} \le c \ \frac{\C_q\z}{(1\vee |\un{e}-\un{e}'|)^d}.
%$$
%Recall from \eqref{relovG} that
%$$
%\na_1 \na_2\ov{G}^e_\mu(\un{e}',\un{e}) = C^e(\un{e}) \Ll(\na_1 \na_2 G^e_\mu(\un{e},\un{e})\Rr) \ \Ll(\na_1 \na_2G\z_\mu(\un{e}',\un{e})\Rr).
%$$
%Since, by \eqref{greenpoint2},
%$$
%\Ll|  C^e(\un{e}) \na_1 \na_2 G^e_\mu(\un{e},\un{e}) \Rr|
%$$
%is bounded uniformly over $\mu$ and $\un{\o}$, the result follows from \eqref{e:sharpgreen}.
%\end{proof}
%\begin{proof}[Proof of Theorem~\ref{t:sharp}]
%We assume that $(\H\z)$ holds. Let $q > 3$. By the assumption, the numbers $\C_q\p$ defined in \eqref{defCqp} are bounded by some constant $\C_q$ uniformly over $p \ge 0$ in a neighbourhood of $0$. From now on, we assume that $p$ is sufficiently small to lie in this neighbourhood. 

%For the beginning of the proof, we can follow the same reasoning as in the proof of Theorem~\ref{t:approxcorr}, up to the point when we arrive at the estimation of the left-hand side of \eqref{e:thestep}, that is,
%$$
%\E\Ll[ \sum_{e\neq e' \in  E\p_\mu}  | \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})| \Rr],
%$$
%where we recall that $E_\mu\p = E\p \cap B_{\mu^{-\gamma}}$, and $\gamma > 1/2$. We rewrite slightly this expectation as
%\begin{equation}
%\label{err1}
%\sum_{e \neq e' \in B_{\mu^{-\gamma}}} \E\Ll[  \1_{e,e' \in E\p} | \na_1 \na_2 G\p_\mu(0,\un{e}')| \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})| \ |\xi + \na \phi\z_\mu(\un{e})| \Rr].
%\end{equation}
%Let $\zeta=1-3/q$. We apply Hölder's inequality with exponents $(\zeta^{-1},q,q,q)$ to bound each summand in \eqref{err1} by
%\begin{multline}
%\label{err2}
%\P[e,e' \in E\p]^{\zeta} \ \E\Ll[ | \na_1 \na_2 G\p_\mu(0,\un{e}')|^{q}\Rr]^{1/q} \\
%\E\Ll[ \ | \na_1 \na_2 G^e_\mu(\un{e}',\un{e})|^{q}\Rr]^{1/q} \ \E\Ll[ |\xi + \na \phi\z_\mu(\un{e})|^{q}\Rr]^{1/q}.
%\end{multline}
%Since we only consider summands for which $e \neq e'$, we have
%$$
%\P[e,e' \in E\p]^{\zeta} = p^{2 \zeta}.
%$$
%By our assumption and Lemma~\ref{l:sharpgreenperturbed}, the product of the second and third terms in \eqref{err2} is bounded by
%$$
%c \ \frac{(\C_q)^2}{(1 \vee |\un{e}'|)^d \ (1 \vee |\un{e} - \un{e}'|)^d},
%$$
%while, by Theorem~\ref{t:approx}, the last term in \eqref{err2} is bounded by some power of $\log(\mu^{-1})$, say $\log^r(\mu^{-1})$. We thus obtain that the sum in \eqref{err1} is bounded by
%$$
%c \ p^{2\zeta} \log^r(\mu^{-1})  \sum_{e \neq e' \in B_{\mu^{-\gamma}}} \frac{(\C_q)^2}{(1 \vee |\un{e}'|)^d \ (1 \vee |\un{e} - \un{e}'|)^d}.
%$$
%By comparing this sum to an integral, we get that the whole expression above is bounded, up to a constant, by 
%$p^{2\zeta} \mu^{-\gamma \eta}$,
%for any $\eta > 0$. Recalling that $\zeta = 1-3/q$ and that $q$ can be chosen arbitrarily large, this concludes the proof.
%\end{proof}
%%
%%
%%
%\begin{proof}[Proof of Theorem~\ref{t:error}] As noted before, it suffices to prove the theorem for $\ov{p} = 0$. The proof follows closely that of Theorem~\ref{t:main0}. Let $\eta > 0$. Instead of \eqref{fixmu}, we fix $\mu = p^2$. With this choice, Theorem~\ref{t:approx} implies that the left-hand side of \eqref{e:approx} is $O(p^{2-\eta})$, while by Theorem~\ref{t:sharp}, the left-hand side of \eqref{e:approx2} is $O(p^{2-3\eta})$. We can then follow the proof of Theorem~\ref{t:main0} without any change until \eqref{e:presque}, whose error term is now $O(p^{2-3\eta})$ instead of $o(p)$. The proof is then concluded in the same way, noting that Theorem~\ref{t:approx} ensures that $\E[\xi \cdot A\z(\xi + \na {\phi}_\mu\z)] = \xi \cdot \Ah\z \xi + O(p^{2-\eta})$.
%\end{proof}






%\begin{proof}[Proof of Theorem~\ref{t:main}]
%The proof follows the same structure as that of Theorem~\ref{t:main0}. Let $\ov{p} \in [0,1)$, and $p \ge 0$ be such that $\ov{p} + p \le 1$. Mimicking the definition in \eqref{deftdphi}, we begin by introducing
%$$
%\td{\phi}_\mu\ppvp = \phi_\mu\pp + \sum_{e \in E\ppp \setminus E\pp} \Ll( \phi_\mu\ppep - \phi_\mu\pp \Rr).
%$$
%Theorem~\ref{t:approxcorr} cannot be repeated to show that there exist constants $c$, $\beta > 0$, $\gamma > 0$ such that for every $p \in [0,1-\ov{p}]$ and every $\mu \in (0,1/2]$,
%\begin{equation}
%\label{e:approxcorrp}
%\E\Ll[\Ll|\na \phi\ppp_\mu - \na \td{\phi}\ppvp_\mu \Rr|\Rr] \le c p^2 \mu^{-1+\beta} + e^{-\mu^{-\gamma}}.
%\end{equation}
%We fix $\mu = p^{1+\eps}$ with $\eps$ satisfying \eqref{e:fixeps}. With the help of \eqref{e:approxcorrp}, we can argue as in the beginning of the proof of Theorem~\ref{t:main0}, and see that $\xi \cdot \Ah\ppp\xi$ is equal, up to an error of order $o(p)$, to
%\begin{multline}
%\label{e:1p}
%\E[\xi \cdot A\ppp(\xi + \na \td{\phi}_\mu\ppvp)] \\
%= \E[\xi \cdot A\ppp(\xi + \na {\phi}_\mu\pp)] + \E\Ll[ \sum_{e \in E\ppp \setminus E\pp} \xi \cdot A\ppp(\underbrace{\na \phi_\mu\ppep - \na \phi_\mu\pp}) \Rr]
%\end{multline}
%(compare with \eqref{e:1}). For every edge $e$ such that $\un{e} = 0$, we define the event $\mfk{E}_p(e)$ by
%$$
%e \in E\ppp \setminus E\pp \quad \text{and} \quad \forall e' \neq e, \un{e}' = 0 \Rightarrow e' \notin E\ppp \setminus E\pp.
%$$
%We can rewrite the first expectation in the right-hand side of \eqref{e:1p} as
%\begin{multline}
%\label{e:2p}
%\E[\xi \cdot A\pp(\xi + \na {\phi}_\mu\pp)] + \\
%\sum_{e: \un{e} = 0}  \E\Ll[\Ll(\underbrace{\xi \cdot A\ppp(\xi + \na {\phi}_\mu\pp) - \xi \cdot A\ppp(\xi + \na {\phi}_\mu\pp)}  \Rr) \1_{\mfk{E}_p(e)}  \Rr] + O(p^2).
%\end{multline}
%On the event $\mfk{E}_p(e)$, the term underbraced above is equal to
%$$
%\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppem) - \xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppem).
%$$
%This quantity is independent of the event $\mfk{E}_p(e)$, whose probability is equal to $p+O(p^2)$. As a consequence, up to an error of order $O(p^2)$, the sum in \eqref{e:2p} is equal to
%$$
%p\sum_{e: \un{e} = 0}  \E\Ll[\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppem) - \xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppem) \Rr].
%$$ 
%We now turn to the second expectation in the right-hand side of \eqref{e:1p}. %We further decompose the sum over the edges according to whether $\un{e} = 0$ or not. 
%For $e \in E\ppp \setminus E\pp$, the quantity underbraced in \eqref{e:1p} is equal to
%$$
%\na \phi_\mu\ppep - \na \phi_\mu\ppem,
%$$
%and this quantity is independent of the event $e \in E\ppp \setminus E\pp$. Arguing as above, we obtain that
%\begin{multline}
%\label{e:3p}
% \E\Ll[ \sum_{e \in E\ppp \setminus E\pp, \un{e} = 0} \xi \cdot A\ppp(\na \phi_\mu\ppep - \na \phi_\mu\pp) \Rr] \\
%=  p \sum_{e : \un{e} = 0} \E\Ll[ \xi \cdot A\ppep(\na \phi_\mu\ppep - \na \phi_\mu\ppem) \Rr] + O(p^2).
%\end{multline}
%For $\un{e} \neq 0$, the quantity $A\ppp$ is also independent of the event $e \in E\ppp \setminus E\pp$. Hence,
%\begin{multline}
%\label{e:4p}
% \E\Ll[ \sum_{e \in E\ppp \setminus E\pp, \un{e} \neq 0} \xi \cdot A\ppp(\na \phi_\mu\ppep - \na \phi_\mu\pp) \Rr] \\
%=  p \sum_{e : \un{e} \neq 0} \E\Ll[ \xi \cdot A\ppp(\na \phi_\mu\ppep - \na \phi_\mu\ppem) \Rr].
%\end{multline}
%Letting $\psi_\mu\pp = \sum_{e \in \B} (\phi_\mu\ppep - \phi_\mu\ppem)$, we can rewrite the latter sum as 
%\begin{equation}
%\label{e:5p}
%\E\Ll[\xi \cdot A\ppp D\psi_\mu\pp\Rr] - \sum_{e : \un{e} = 0} \E\Ll[ \xi \cdot A\ppp(\na \phi_\mu\ppep - \na \phi_\mu\ppem) \Rr].
%\end{equation}
%As for the second term, we have
%\begin{multline}
%\label{e:6p}
%\sum_{e : \un{e} = 0} \E\Ll[ \xi \cdot A\ppp(\na \phi_\mu\ppep - \na \phi_\mu\ppem) \Rr] \\
%= \sum_{e : \un{e} = 0} \E\Ll[ \xi \cdot A\pp(\na \phi_\mu\ppep - \na \phi_\mu\ppem) \Rr] + O(p).
%\end{multline}
%For the first term in \eqref{e:5p}, following the argument of Proposition~\ref{p:defcoef}, we first remark that $D\psi_\mu\pp$ converges in $L^2(\un{\O})$ as $\mu$ tends to $0$. %, to some $\chi\pp \in L^2_\na$. 
%Hence, we have
%$$
%\E\Ll[\xi \cdot A\ppp D\psi_\mu\pp\Rr] = \E\Ll[\xi \cdot A\pp D\psi_\mu\pp\Rr] + O(p),
%$$
%and moreover, $\E[\xi \cdot A\pp D\psi_\mu\pp]$ converges to some constant as $\mu$ tends to $0$. 
%
%To sum up, we have shown that
%$$
%\xi \cdot \Ah\ppp \xi = \E[\xi \cdot A\pp(\xi + \na {\phi}_\mu\pp)] + p \ a_1\ppp(\mu) + o(p),
%$$
%where
%\begin{multline*}
%a_1\pp(\mu) = \sum_{e: \un{e} = 0}  \E\Ll[\xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppep) - \xi \cdot A\ppep(\xi + \na {\phi}_\mu\ppem) \Rr] \\
% + \E\Ll[\xi \cdot A\pp D\psi_\mu\pp\Rr] -  \sum_{e : \un{e} = 0} \E\Ll[ \xi \cdot A\pp(\na \phi_\mu\ppep - \na \phi_\mu\ppem) \Rr].
%\end{multline*}
%Using the definition of $\psi_\mu$, one can check that $a_1\pp(\mu)$ indeed satisfies \eqref{defa1mupp}. By Theorem~\ref{t:approx} and the fact that we fixed $\mu = p^{1+\eps}$, we have $\E[\xi \cdot A\pp(\xi + \na {\phi}_\mu\pp)] = \xi \cdot \Ah\pp \xi + o(p)$. There remains to show that $a_1\pp(\mu)$ converges to a constant as $\mu$ tends to $0$. We have already seen that $\E[\xi \cdot A\pp D\psi_\mu\pp]$ converges to a constant. The fact that $\na {\phi}_\mu\ppep(0,\cdot)$ and $\na {\phi}_\mu\ppem(0,\cdot)$ both converge in $L^2(\un{\O})$ as $\mu$ tends to $0$ is obtained as in Proposition~\ref{p:corrlim}.
%
%We have thus justified the expansion announced in \eqref{e:main}, assuming $\ov{p} \in [0,1)$ and $p \ge 0$. If we assume instead that $\ov{p} \in (0,1]$ and $p \le 0$, then we can follow the same reasoning as above, and arrive at the same formula \eqref{e:main}. Hence, the proof is complete.
%\end{proof}



\bigskip

\noindent \textbf{Acknowledgements.} I would like to warmly thank Vincent Beffara, Noam Berger, Antoine Gloria, Claude Le Bris and Frédéric Legoll for stimulating discussions about this problem.




\end{document}








